ROBERT PARRINO . DAVID KIDWELL . THOMAS BATES

FUNDAMENTALS OF CORPORATE FINANCE

THIRD EDITION

WILEY

Selected Abbreviations and Notation

$\beta =$	beta (a measure of systematic risk)
$\Delta =$	change (e.g., ΔP = change in price level, ΔS = change in sales level)
$\rho =$	correlation
$\sigma^{2}(\sigma) =$	variance (standard deviation)
x =	fractional weight of investment or component of capital
Add WC =	addition to working capital
APR =	annual percentage rate
ARR =	accounting rate of return
b =	dividend payout ratio
C =	coupon payment (bond), call option value
Cap Exp =	capital expenditures
CF =	cash flow
CF Opns =	cash flow from operations
CFI =	cash flow to investors
CFLTA =	cash flow invested in long-term assets
CFNWC =	cash flow invested in net working capital
CFOA =	cash flow from operating activity
CIP =	call interest premium
CO =	crossover level of unit sales
COGS =	cost of goods sold
CV =	coefficient of variation
D =	dividend (stock)
D&A =	depreciation and amortization
DOL =	degree of operating leverage
DPO =	days' payables outstanding
DRP =	default risk premium
DSI =	days' sales in inventory
DSO =	days' sales outstanding
$E(\bullet) =$	expected value ($E(R)$ = expected return, etc.)
EAC =	equivalent annual cost
EAR = EAY =	effective annual rate (yield)
EBIT =	earnings before interest and taxes
EBITDA =	earnings before interest, taxes, depreciation, and amortization
EBT =	earnings before taxes
EFN =	external funding needed
EOQ =	economic order quantity
EROA =	EBIT return on assets
F =	face value (bond)
FC =	fixed costs
FCF =	free cash flows
FCFE =	free cash flow to equity
FCFF =	free cash flow from the firm

FV =	future value
$FVA_n =$	future value of an annuity
FXR =	foreign exchange or currency risk premium
<i>g</i> =	growth rate
<i>i</i> =	nominal rate of interest
IGR =	internal growth rate
IRR =	internal rate of return
k =	cost of capital (debt or equity)
m =	number of payments per year
MAT =	maturity adjustment to cost of a loan
MRP =	marketability risk premium
MV =	market value
n =	number of periods
NCF =	net cash flow
NCFOA =	net cash flow from operating activities
NOPAT =	net operating profits after tax
NPV =	net present value
NWC =	net working capital
OC =	operating cycle
Op Ex =	cash operating expenses
p =	probability
P =	price (P_0 = price at time zero, etc.), put option value
P/E ratio =	price/earnings ratio
PB =	payback period
PI =	profitability index
PR =	prime rate
PV =	present value
PV annuity factor =	present value of annuity factor
$PVA_n =$	present value of an annuity
PVP =	present value of a perpetuity
r =	real rate of interest
R =	return ($R_{rf} = risk$ free, R_i , $R_{Portfolio}$, etc.)
ROA =	return on assets
ROE =	return on equity
S =	Sharpe Ratio
SGR =	sustainable growth rate
t =	tax rate
TV =	terminal value
V =	value (e.g., $V_{Firm} = V_{Assets} = V_{Debt} + V_{Equity}$)
VC =	variable costs
WACC =	weighted average cost of capital

WileyPLUS

Now with: ORION, An Adaptive Experience

WileyPLUS is a research-based online environment for effective teaching and learning.

WileyPLUS builds students' confidence because it takes the guesswork out of studying by providing students with a clear roadmap:

- what to do
- how to do it
- if they did it right

It offers interactive resources along with a complete digital textbook that help students learn more. With *WileyPLUS*, students take more initiative so you'll have greater impact on their achievement in the classroom and beyond.



Blackboard

For more information, visit www.wileyplus.com

WileyPLUS

ALL THE HELP, RESOURCES, AND PERSONAL SUPPORT YOU AND YOUR STUDENTS NEED! www.wileyplus.com/resources



2-Minute Tutorials and all of the resources you and your students need to get started

WileyPLUS

Student Partner Program

Student support from an experienced student user

Wiley Faculty Network



Collaborate with your colleagues, find a mentor, attend virtual and live events, and view resources www.WhereFacultyConnect.com



Pre-loaded, ready-to-use assignments and presentations created by subject matter experts



Technical Support 24/7 FAQs, online chat, and phone support www.wileyplus.com/support



Your *WileyPLUS* Account Manager, providing personal training and support

FUNDAMENTALS OF CORPORATE FINANCE

Robert Parrino

Department Chair and Lamar Savings Centennial Professor of Finance University of Texas at Austin

David S. Kidwell

Professor of Finance and Dean Emeritus University of Minnesota

Thomas W. Bates

Department Chair and Associate Professor of Finance Arizona State University



VICE PRESIDENT, EXECUTIVE PUBLISHER	George Hoffman
EXECUTIVE EDITOR	Joel Hollenbeck
PROJECT EDITOR	Jennifer Manias
ASSISTANT EDITOR	Courtney Luzzi
SENIOR EDITORIAL ASSISTANT	Tai Harriss
DIRECTOR OF MARKETING	Amy Scholz
SENIOR MARKETING MANAGER	Karolina Zarychta Honsa
MARKETING ASSISTANT	Mia Brady
SENIOR PRODUCT DESIGNER	Allie K. Morris
PRODUCT DESIGNER	Greg Chaput
SENIOR MEDIA SPECIALIST	Elena Santa Maria
CONTENT MANAGER	Dorothy Sinclair
SENIOR PRODUCTION EDITOR	Valerie A. Vargas
SENIOR PHOTO EDITOR	MaryAnn Price
PHOTO RESEARCHER	Amanda Bustard
DESIGN DIRECTOR	Harry Nolan
SENIOR DESIGNER	Maureen Eide
COVER PHOTO	© Sironpe/iStockphoto

COVER PHOTO © Sironpe/iStockphoto The CFA Institute Materials used in this book are reproduced and republished from the CFA Program Materials with permission from the CFA Institute. The authors and publisher are grateful for permission to use this material.

This book was set in Minion Pro by Aptara, Inc. and printed and bound by Courier/Kendallville. The cover was printed by Courier/Kendallville.

This book is printed on acid free paper. ∞

Founded in 1807, John Wiley & Sons, Inc. has been a valued source of knowledge and understanding for more than 200 years, helping people around the world meet their needs and fulfill their aspirations. Our company is built on a foundation of principles that include responsibility to the communities we serve and where we live and work. In 2008, we launched a Corporate Citizenship Initiative, a global effort to address the environmental, social, economic, and ethical challenges we face in our business. Among the issues we are addressing are carbon impact, paper specifications and procurement, ethical conduct within our business and among our vendors, and community and charitable support. For more information, please visit our website: www.wiley.com/go/citizenship.

Copyright © 2015, 2012, 2009 John Wiley & Sons, Inc. All rights reserved. No part of this publication may be reproduced, stored in a retrieval system, or transmitted in any form or by any means, electronic, mechanical, photocopying, recording, scanning or otherwise, except as permitted under Sections 107 or 108 of the 1976 United States Copyright Act, without either the prior written permission of the Publisher, or authorization through payment of the appropriate per-copy fee to the Copyright Clearance Center, Inc., 222 Rosewood Drive, Danvers, MA 01923 (website: www.copyright.com). Requests to the Publisher for permission should be addressed to the Permissions Department, John Wiley & Sons, Inc., 111 River Street, Hoboken, NJ 07030-5774, (201) 748-6011, fax (201) 748-6008, or online at: www.wiley.com/go/permissions.

Evaluation copies are provided to qualified academics and professionals for review purposes only, for use in their courses during the next academic year. These copies are licensed and may not be sold or transferred to a third party. Upon completion of the review period, please return the evaluation copy to Wiley. Return instructions and a free of charge return mailing label are available at www.wiley.com/go/returnlabel. If you have chosen to adopt this textbook for use in your course, please accept this book as your complimentary desk copy. Outside of the United States, please contact your local sales representative.

ISBN: 978-1-118-84589-9 BRV ISBN: 978-1-118-90166-3

Printed in the United States of America

 $10\ 9\ 8\ 7\ 6\ 5\ 4\ 3\ 2\ 1$

Dedication

ROBERT PARRINO

To my parents, whose life-long support and commitment to education inspired me to become an educator, and to my wife, Emily, for her unending support.

DAVID KIDWELL

To my parents, Dr. William and Margaret Kidwell, for their endless support of my endeavors; to my son, David Jr., of whom I am very proud; and to my wife, Jillinda, who is the joy of my life.

THOMAS BATES

To my wife, Emi, and our daughters, Abigail and Lillian. Your support, patience, fun, and friendship make me a better educator, scholar, and person.



ROBERT PARRINO

Department Chair and Lamar Savings Centennial Professor of Finance Director, Hicks, Muse, Tate & Furst Center for Private Equity Finance McCombs School of Business, University of Texas at Austin

A member of the faculty at University of Texas since 1992, Dr. Parrino teaches courses in regular degree and executive education programs at the University of Texas, as well as in customized executive education courses for industrial, financial, and professional firms. He has also taught at the University of Chicago, University of Rochester, and IMADEC University in Vienna. Dr. Parrino has received numerous awards for teaching excellence at the University of Texas from students, faculty, and the Texas Exes (alumni association).

Dr. Parrino has been involved in advancing financial education outside of the classroom in a variety of ways. As a Chartered Financial Analyst (CFA) charterholder, he has been very active with the CFA Institute, having been a member of the candidate curriculum committee, served as a regular speaker at the annual Financial Analysts Seminar, spoken at over 20 Financial Analyst Society meetings, and served as a member of the planning committee for the CFA Institute's Annual Meeting. In addition, Dr. Parrino is the founding director of the Hicks, Muse, Tate & Furst Center for Private Equity Finance at the University of Texas. The center sponsors conferences and other educational activities in areas related to private equity finance. Dr. Parrino was Vice President for Financial Education of the Financial Management Association (FMA) from 2008 to 2010 and an academic director of the FMA from 2011 to 2013. Since 2009 he has served as a member of the Scientific Panel for the Center for Corporate Investor Responsibility in the Sim Kee Boon Institute of Financial Economics at Singapore Management University.

Dr. Parrino also co-founded the Financial Research Association and is Associate Editor of the *Journal of Corporate Finance*. Dr. Parrino's research focus includes corporate governance, financial policies, restructuring, mergers and acquisitions, and private equity markets. He has published his research in a number of journals, including the *Journal of Finance, Journal of Financial Economics, Journal of Financial and Quantitative Analysis, Journal of Law and Economics, Journal of Portfolio Management*, and *Financial Management*. Dr. Parrino has won a number of awards for his research, including the 2013–2014 Career Award for Outstanding Research Contributions at the McCombs School of Business.

Dr. Parrino has experience in the application of corporate finance concepts in a variety of business situations. Since entering the academic profession, he has been retained as an advisor on valuation issues concerning businesses with enterprise values ranging to more than \$1 billion and has consulted in areas such as corporate financing, compensation, and corporate governance. Dr. Parrino was previously President of Sprigg Lane Financial, Inc., a financial consulting firm with offices in Charlottesville, Virginia, and New York City. While at Sprigg Lane, he was on the executive, banking, and portfolio committees of the holding company that owns Sprigg Lane. Before joining Sprigg Lane, Dr. Parrino was on the Corporate Business Planning and Development staff at Marriott Corporation. At Marriott, he conducted fundamental business analyses and preliminary financial valuations of new business development opportunities and potential acquisitions. Dr. Parrino holds a B.S. in chemical engineering from Lehigh University, an MBA degree from The College of William and Mary, and M.S. and Ph.D. degrees in applied economics and finance, respectively, from the University of Rochester.



DAVID S. KIDWELL

Professor of Finance and Dean Emeritus Curtis L. Carlson School of Management, University of Minnesota

Dr. Kidwell has over 30 years experience in financial education, as a teacher, researcher, and administrator. He has served as Dean of the Carlson School at the University of Minnesota and of the School of Business Administration at the University of Connecticut. Prior to joining the University of Connecticut, Dr. Kidwell held endowed chairs in banking and finance at Tulane University, the University of Tennessee, and Texas Tech University. He was also on the faculty at the Krannert Graduate School of Management, Purdue University where he was twice voted the outstanding undergraduate teacher of the year.

An expert on the U.S. financial system, Dr. Kidwell is the author of more than 80 articles dealing with the U.S. financial system and capital markets. He has published his research in the leading journals, including *Journal of Finance, Journal of Financial Economics, Journal of Financial and Quantitative Analysis, Financial Management*, and *Journal of Money, Credit, and Banking*. Dr. Kidwell has also participated in a number of research grants funded by the National Science Foundation to study the efficiency of U.S. capital markets, and to study the impact of government regulations upon the delivery of consumer financial services.

Dr. Kidwell has been a management consultant for Coopers & Lybrand and a sales engineer for Bethlehem Steel Corporation. He served on the Board of Directors for the Schwan Food Company and was the Chairman of the Audit and Risk Committee. Dr. Kidwell is the past Secretary-Treasurer of the Board of Directors of AACSB, the International Association for Management Education and is a past member of the Boards of the Minnesota Council for Quality, the Stonier Graduate School of Banking, and Minnesota Center for Corporate Responsibility. Dr. Kidwell has also served as an Examiner for the 1995 Malcolm Baldrige National Quality Award, on the Board of Directors of the Juran Center for Leadership in Quality, and on the Board of the Minnesota Life Insurance Company.

Dr. Kidwell holds an undergraduate degree in mechanical engineering from California State University at San Diego, an MBA with a concentration in finance from California State University at San Francisco, and a Ph.D. in finance from the University of Oregon.



THOMAS W. BATES Department Chair and Associate Professor of Finance W. P. Carey School of Business, Arizona State University

Dr. Bates is the Chair of the Department of Finance and Dean's Council of 100 Distinguished Scholar at the W. P. Carey School of Business, Arizona State University. He has also taught courses in finance at the University of Delaware, the Ivey School of Business at the University of Western Ontario, and the University of Arizona where he received the Scrivner teaching award. During his career as an educator, Professor Bates has taught corporate finance to students in undergraduate, MBA, executive MBA, and Ph.D. programs, as well as in custom corporate educational courses.

Professor Bates is a regular contributor to the academic finance literature in such journals as *The Journal of Finance*, *Journal of Financial Economics*, and *Financial Management*. His research addresses a variety of issues in corporate finance including the contracting environment in mergers and acquisitions, corporate liquidity decisions and cash holdings, and the governance of corporations. In practice, Dr. Bates has worked with companies and legal firms as an advisor on issues related to the valuation of companies and corporate governance. Dr. Bates received a B.A. in Economics from Guilford College and his doctorate in finance from the University of Pittsburgh.

Preface

We have written *Fundamentals of Corporate Finance* for use in an introductory course in corporate finance at the undergraduate level. It is also suitable for advanced undergraduate, executive development, and traditional or executive MBA courses when supplemented with cases and outside readings. The main chapters in the book assume that students are well-versed in algebra and that they have taken courses in principles of economics and financial accounting. Optional chapters covering important economic and financial accounting concepts are included for students and instructors seeking such coverage.

Balance Between Conceptual Understanding and Computational Skills

We wrote this corporate finance text for one very important reason. We want to provide students and instructors with a book that strikes the best possible balance between helping students develop an intuitive understanding of key financial concepts and providing them with problem-solving and decision-making skills. In our experience, teaching students at all levels and across a range of business schools, we have found that students who understand the intuition underlying the basic concepts of finance are better able to develop the critical judgment necessary to apply financial tools to a broad range of real-world situations. An introductory corporate finance course should provide students with a strong understanding of both the concepts and tools that will help them in their subsequent business studies and their personal and professional lives.

Market research supports our view. Many faculty members who teach the introductory corporate finance course to undergraduates want a book that bridges the gap between conceptually-focused and computationally-focused books. This text is designed to bridge this gap. Specifically, it develops the fundamental concepts underlying corporate finance in an intuitive manner while maintaining a strong emphasis on developing computational skills. This text also takes the students one step further by emphasizing the use of intuition and analytical skills in decision making.

Our ultimate goal has been to write a book and develop associated learning tools that help our colleagues succeed in the classroom—materials that are genuinely helpful in the learning process. Our book offers a level of rigor that is appropriate for finance majors and yet presents the content in a manner that both finance and non-finance students find accessible and want to read. Writing a book that is both *rigorous* and *accessible* has been one of our key objectives, and both faculty and student reviews of the first and second editions suggest that we have achieved this objective.

We have also tried to provide solutions to many of the challenges facing finance faculty in the current environment, who are asked to teach ever-increasing numbers of students with limited resources. Faculty members need a book and associated learning tools that help them effectively leverage their time. The organization of this book and the supplemental materials, along with the innovative *WileyPLUS* Web-based interface, which offers extensive problem solving opportunities and other resources for students, provide such leverage to an extent not found with other textbooks.

A Focus on Value Creation

This book is more than a collection of ideas, equations, and chapters. It has an important integrating theme—that of value creation. This theme, which is carried throughout the book, provides a framework that helps students understand the relations between the various concepts covered in the book and makes it easier for them to learn these concepts.

The concept of value creation is the most fundamental notion in corporate finance. It is in stockholders' best interests for value maximization to be at the heart of the financial decisions made within the firm. Thus, it is critical that students be able to analyze and make business decisions with a focus on value creation. The concept of value creation is introduced in the first chapter of the book and is further developed and applied throughout the remaining chapters.

The theme of value creation is operationalized through the net present value (NPV) concept. Once students grasp the fundamental idea that financial decision makers should only choose courses of action whose benefits exceed their costs, analysis and decision making using the NPV concept becomes second nature. By helping students better understand the economic rationale for a decision from the outset, rather than initially focusing on computational skills, our text keeps students focused on the true purpose of the calculations and the decision at hand.

Integrated Approach: Intuition, Analysis, and Decision Making

To support the focus on value creation, we have emphasized three things: (1) providing an intuitive framework for understanding fundamental finance concepts, (2) teaching students how to analyze and solve finance problems, and (3) helping students develop the ability to use the results from their analyses to make good financial decisions.

- 1. An Intuitive Approach: We believe that explaining finance concepts in an intuitive context helps students develop a richer understanding of those concepts and gain better insights into how finance problems can be approached. It is our experience that students who have a strong conceptual understanding of financial theory better understand how things really work and are better problem solvers and decision makers than students who focus primarily on computational skills.
- 2. Analysis and Problem Solving: With a strong understanding of the basic principles of finance, students are equipped to tackle a wide range of financial problems. In addition to the many numerical examples that are solved in the text of each chapter, this book has almost 1,200 end-of-chapter homework and review problems that have been written with Bloom's Taxonomy in mind. Solutions for these problems are provided in the Instructor's Manual. We strive to help students acquire the ability to analyze and solve finance problems.
- **3. Decision Making:** In the end, we want to prepare students to make sound financial decisions. To help students develop these skills, throughout the text we illustrate how the results from financial analyses are used in decision making.

Organization and Coverage

In order to help students develop the skills necessary to tackle investment and financing decisions, we have arranged the book's 21 chapters into five major building blocks, that collectively comprise the seven parts of the book, as illustrated in the accompanying exhibit and described below.

Introduction

Part 1, which consists of Chapter 1, provides an introduction to corporate finance. It describes the role of the financial manager, the types of fundamental decisions that financial mangers make, alternative forms of business organization, the goal of the firm, agency conflicts and how they arise, and the importance of ethics in financial decision-making. These discussions set the stage and provide a framework that students can use to think about key concepts as the course progresses.

Foundations

Part 2 of the text consists of Chapters 2 through 4. These chapters present the basic institutional, economic, and accounting knowledge and tools that students should understand before they begin the study of financial concepts. Most of the material in these chapters is typically taught in other courses. Since students come to the corporate finance course with varying academic backgrounds, and because the time that has elapsed since students have taken particular prerequisite courses also varies, the chapters in Part 2 can help the instructor ensure that all students have the same base level of knowledge early in the course. Depending on the educational background of the students, the instructor might not find it necessary to cover all or any of the material in these chapters. Some or all of these chapters might, instead, be assigned as supplemental readings.

Chapter 2 describes the services financial institutions provide to businesses, how domestic and international financial markets work, the concept of market efficiency, how firms use financial markets, and how interest rates are determined in the economy. Chapter 3 describes the key financial statements and how they are related, as well as how these statements are related to cash flows to investors. Chapter 4 discusses ratio analysis



and other tools used to evaluate financial statements. Throughout Part 2, we emphasize the importance of cash flows to get students thinking about cash flows as a critical component of all valuation calculations and financial decisions.

Basic Concepts and Tools

Part 3 presents basic financial concepts and tools and illustrates their application. This part of the text, which consists of Chapters 5 through 9, introduces time value of money and risk and return concepts and then applies these concepts to bond and stock valuation. These chapters provide students with basic financial intuitions and computational tools that will serve as the building blocks for analyzing investment and financing decisions in subsequent chapters.

Analysis

Parts 4 and 5 of the text focus on investment and financing decisions. Part 4 covers capital budgeting. Chapter 10 introduces the concept of net present value and illustrates its application as the principle tool for evaluating capital projects. It also discusses alternative capital budgeting decision rules, such as internal rate of return, payback period, and accounting rate of return, and compares them with the net present value criterion. Finally, Chapter 10 also discusses investment decisions with capital rationing. The discussions in Chapter 10 provide a framework that will help students in the rest of Part 4 as they learn the nuances of capital budgeting analysis in realistic settings.

Chapters 11 and 12 follow with in-depth discussions of how cash flows are calculated and forecast. The cash flow calculations are presented in Chapter 11 using a valuation framework that helps students think about valuation concepts in an intuitive way and that prepares them for the extension of these concepts to business valuation in Chapter 18. Chapter 12 covers analytical tools—such as breakeven, sensitivity, scenario, and simulation analysis—that give students a better appreciation for how they can deal with the uncertainties associated with cash flow forecasts.

Chapter 13 explains how the discount rates used in capital budgeting are estimated. This chapter uses an innovative concept—that of the finance balance sheet—to help students develop an intuitive understanding of the relations between the costs of the individual components of capital and the firm's overall weighted average cost of capital. It also provides a detailed discussion of methods used to estimate the costs of the individual components of capital that are used to finance a firm's investments and how these estimates are used in capital budgeting.

Part 5 covers working capital management and financing decisions. It begins, in Chapter 14, with an introduction to how firms manage their working capital and the implications of working capital management decisions for financing decisions and firm value. This is followed, in Chapters 15 and 16, with discussions of how firms raise capital to fund their real activities and the factors that affect how firms choose among the various sources of capital available to them. Chapter 16 also includes an extensive appendix on leasing concepts and buy vs. lease analysis. Chapter 17 rounds out the discussion of financing decisions with an introduction to dividends, stock repurchases, stock dividends and splits, and payout policy.

Integration

Part 6, which consists of Chapters 18 and 19, brings together many of the key concepts introduced in the earlier parts of the text. Chapter 18 covers financial aspects of business formation and growth and introduces students to business valuation concepts for both private and public firms. The discussions in this chapter integrate the investment and financing concepts discussed in Parts 4 and 5 to provide students with a more complete picture of how all the financial concepts fit together. Chapter 19 covers concepts related to financial planning, forecasting, and managing growth.

Part 7 introduces students to some important issues that managers must deal with in applying the concepts covered in the text to real-world problems. Chapter 20 introduces call and put options and discusses how they relate to investment and financing decisions. It describes options that are embedded in the securities that firms issue. It also explains, at an accessible level, the idea behind real options and why traditional NPV analysis does not take such options into account. In addition, the chapter discusses agency costs of debt and equity and the implications of these costs for investment and financing decisions. Finally, Chapter 20 illustrates the use of options in risk management. Instructors can cover the topics in Chapter 20 near the end of the course or insert them at the appropriate points in Parts 4 and 5. Chapter 21 examines how international considerations affect the application of concepts covered in the book.

Unique Chapters

Chapter on Business Formation, Growth, and Valuation

We wrote Chapter 18 in response to students' heightened interest in new business formation (entrepreneurship) and in order to draw together, in a comprehensive way, the key concepts from capital budgeting, working capital management, and financial policy. This capstone chapter provides an overview of practical finance issues associated with forecasting cash flows and capital requirements for a new business, preparing a business plan, and business valuation. The discussion of business valuation extends far beyond that found in other introductory corporate finance textbooks.

Chapter on Options and Corporate Finance

Many other corporate finance textbooks have a chapter that introduces students to financial options and how they are valued. This chapter goes further. It provides a focused discussion of the different types of financial and non-financial options that are of concern to financial managers, including options embedded in debt and equity securities, real options and their effect on project analysis, how option-like payoff functions faced by stockholders, bondholders, and managers affect agency relationships, and the use of options in risk management.

Proven Pedagogical Framework

We have developed several distinctive features throughout the book to aid student learning. The pedagogical features included in our text are as follows:



David Faul Worns/Bloomberg via Getty Image

hen Blockbuster Inc. filed for bankruptcy protection in September 2010, it looked like Netflix was unstoppable. With the demise of its major competitor in the video rental market, Netflix had become the most successful company in its industry. On the day of the Blockbuster filing, Netflix's stock price closed at \$160.47 per share, up from \$46.85 just a year earlier. By early July of the following year, its shares were trading for as much as \$304.79.

Then Netflix managers made a decision that did not turn out as expected. On July 12, 2011, they announced a new pricing strategy that would raise prices by as much as 60 percent for millions of Netflix subscribers who wanted to both rent DVDs by mail and watch video on the internet. Those two services would be unbundled and, instead of paying \$10 per month for both, customers would have to buy them separately at a cost of at least \$16 per month.

Customers and investors reacted swiftly and negatively to this announcement. During the fiscal quarter from July to September 2011, Netflix lost 800,000 U.S. subscribers. Although Netflix managers did announce in October that they were reversing their decision to unbundle the DVD rentals and video streaming, the damage had been done. The company's stock price continued to drift down, finishing the year at 570 per share.

Throughout 2012, Netflix's stock price fluctuated between \$53.80 and \$129.25 per share and it looked unlikely that the share price would reach \$300 again anytime soon. However, on January 23, 2013, Netflix surprised investors by announcing quarterly earnings of \$0.13 per share when analysts were expecting a loss of about that amount. Between January 23 and January 25 of 2013, Netflix's share price jumped from \$103.26 to \$169.56. Subsequent good news about Netflix and a general increase in stock market prices pushed the share price to \$366.31 by December 16, 2013.

Netflix shares were considered a risky investment even at the time of the Blockbuster bankruptcy filing, yet few investors would have thought that the price of those shares would change as quickly and by as much as it did. An investor who purchased Netflix shares at \$300 in July 2011 and sold at \$70 on December 2011

Learning Objectives

Explain the relation between risk and return. Describe the two components of a total holding period return, and calculate this return for an asset.

calculate the expected return for an asset. Explain what the standard deviation of

- returns is and why it is very useful in finar and calculate it for an asset.
- Explain the concept of diversificatio
- Discuss which type of risk ma
- Describe what the Capital Asset Pricing Model (CAPM) tells us and how to use it to evaluate whether the expected return of an asset is sufficient to compensate an investor

199

CHAPTER SEVEN

CHAPTER OPENER VIGNETTES

Each chapter begins with a vignette that describes a real company or personal application. The vignettes illustrate concepts that will be presented in the chapter and are meant to heighten student interest, motivate learning, and demonstrate the real-life relevance of the material in the chapter.

LEARNING OBJECTIVES

The opening vignette is accompanied by learning objectives that identify the most important material for students to understand while reading the chapter. At the end of the chapter, the Summary of Learning Objectives summarizes the chapter content in the context of the learning objectives.



Calculating the Return on an Investment

PROBLEM: You purchased a beat-up 1974 Datsun 240Z sports car a year ago for \$1,500. Datsun is what Nissan, the Japanese car company, was called in the 1970s. The 240Z was the first in a series of cars that led to the Nissan 370Z that is being sold today. Recognizing that a mint-condition 240Z is a much sought-after car, you invested \$7,000 and a lot of your time fixing up the car. Last week, you sold it to a collector for \$18,000. Not counting the value of the time you spent restoring the car, what is the total return you earned on this investment over the one-year holding period?

APPROACH: Use Equation 7.1 to calculate the total holding period return. To calculate R_T using Equation 7.1, you must know P_0 , P_1 , and CF_1 . In this problem, you can assume that the \$7,000 was spent at the time you bought the car to purchase parts and materials. Therefore, your initial investment, P_0 , was \$1,500 + \$7,000 = \$8,500. Since there were no other cash inflows or outflows between the time that you bought the car and the time that you sold it, CF_1 equals \$0.

SOLUTION: The total holding period return is:

 $R_T = R_{CA} + R_I = \frac{P_1 - P_0 + CF_1}{P_0} = \frac{\$18,000 - \$8,500 + \$0}{\$8,500} = 1.118, \text{ or } 111.8\%$

.....LEARNING BY DOING APPLICATIONS

Along with a generous number of in-text examples, most chapters include several Learning by Doing Applications. These applications contain quantitative problems with step-by-step solutions to help students better understand how to apply their intuition and analytical skills to solve important problems. By including these exercises, we provide students with additional practice in the application of the concepts, tools, and methods that are discussed in the text.

BUILDING INTUITION

Students must have an intuitive understanding of a number of important principles and concepts to successfully master the finance curriculum. Throughout the book, we emphasize these important concepts by presenting them in Building Intuition boxes. These boxes provide a statement of an important finance concept, such as the relation between risk and expected return, along with an intuitive example or explanation to help the student "get" the concept. These boxes help the students develop finance intuition. Collectively the Building Intuition boxes cover the most important concepts in corporate finance.

MORE RISK MEANS A HIGHER EXPECTED RETURN The greater the risk associated with an invest-

ment, the greater the return investors expect from

it. A corollary to this idea is that investors want the highest return for a given level of risk or the lowest risk for a given level of return. When choosing between two investments that have the same level of risk, investors prefer the investment with the higher return. Alternatively, if two investments have the same expected return, investors prefer the less risky alternative.



DECISION-MAKING EXAMPLES

Throughout the book, we emphasize the role of the financial manager as a decision maker. To that end, twenty chapters include Decision-Making Examples. These examples, which emphasize the decision-making process rather than computation, provide students with experience in financial decision making. Each Decision-Making Example outlines a scenario and asks the student to make a decision based on the information presented.

END OF CHAPTER PEDAGOGY

SUMMARY OF LEARNING OBJECTIVES AND KEY EQUATIONS

At the end of the chapter, you will find a summary of the key chapter content related to each of the learning objectives listed at the beginning of the chapter, as well as an exhibit listing the key equations in the chapter.

SUMMARY OF Learning Objectives

- Explain the relation between risk and return. Investors require higher returns for taking greater risk. They prefer the investment with the highest possible return for a given level of risk or the investment with the lowest risk for a given level of return.
- 2 Describe the two components of a total holding period return, and calculate this return for an asset. The total holding period return on an investment consists of a capital appreciation component and an income component. This return is calculated using Equation 7.1. It is important to recognize that investors do not care whether they receive a dollar of return through capital appreciation or as a cash dividend. Investors value both sources of return equally.
- 3 Explain what an expected return is and calculate the expected return for an asset.

The expected return is a weighted average of the possible returns from an investment, where each of these returns is weighted by the

5 Explain the concept of diversification.

Diversification entails reducing risk by investing in two or more assets whose values do not always move in the same direction at the same time. Investing in a portfolic containing assets whose prices do not always move together reduces risk because some of the changes in the prices of individual assets offset each other. This can cause the overall variance in the returns of an investor's portfolio to be lower than if it consisted of only a single asset.

- O Discuss which type of risk matters to investors and why. Investors care about only systematic risk. This is because they can eliminate unsystematic risk by holding a diversified portfolio. Diversified investors will bid up prices for assets to the point at which they are just being compensated for the systematic risks they must bear.
 - Describe what the Capital Asset Pricing Model (CAPM) tells us and how to use it to evaluate whether the expected return of an asset is sufficient to compensate an investor for the risks associated with that asset.

• SUMMARY OF Key Equations

Equation	Description	Formula
7.1	Total holding period return	$R_{T} = R_{CA} + R_{I} = \frac{P_{1} - P_{0}}{P_{0}} + \frac{CF_{1}}{P_{0}} = \frac{\Delta P + CF_{1}}{P_{0}}$
7.2	Expected return on an asset	$E(R_{Asset}) = \sum_{i=1}^{n} (p_i \times R_i)$
7.3	Variance of return on an asset	$Var(R) = \sigma_R^2 = \sum_{i=1}^{n} \{p_i \times [R_i - E(R)^2]\}$



SELF-STUDY PROBLEMS WITH SOLUTIONS

Five problems similar to the in-text Learning by Doing Applications follow the summary and provide additional examples with step-by-step solutions to help students further develop their problem-solving and computational skills.

DISCUSSION QUESTIONS

At least ten qualitative questions, called Discussion Questions, require students to think through their understanding of key concepts and apply those concepts to a problem.

Discussion Questions

- 7.1 Suppose that you know the risk and the expected return for two stocks. Discuss the process you might utilize to determine which of the two stocks is a better buy. You may assume that the two stocks will be the only assets held in your portfolio.
- 7.2 What is the difference between the expected rate of return and the required rate of return? What does it mean if they are different for a particular asset at a particular point in time?
- 7.3 Suppose that the standard deviation of the returns on the shares of stock at two different companies is exactly the same. Does this mean that the required rate of return will be the same for these two stocks? How might the required rate of return on the stock of a third company be greater than the required rates of return on the stocks of the first two companies even if the standard deviation of the returns of the third company's stock is lower?
- 7.4 The correlation between stocks A and B is 0.50, while the correlation between stocks A and C is -0.5. You already own stock A and are thinking of buying either stock B or stock C. If you want your portfolio to have the lowest possible risk, would you buy stock B or C? Would you expect the stock you choose to affect the return that you earn on your portfolio?
 7.5 The idea that we can know the return on a security for each possible outcome is overly simplistic.
- 7.5 The idea that we can know the return on a security for each possible outcome is overly simplistic. However, even though we cannot possibly predict all possible outcomes, this fact has little bearing on the risk-free return. Explain why.

Questions and Problems

QUESTIONS AND PROBLEMS

The Questions and Problems, numbering 26 to 48 per chapter, are primarily quantitative and are classified as Basic, Intermediate, or Advanced.

BASIC 7.1 Returns: Describe the difference between a total holding period return and an expected return. 7.2 Expected returns: John is watching an old game show rerun on television called *Let's Make a Deal* in which the contestant chooses a prize behind one of two curtains. Behind one of the curtains is a gag prize worth \$150, and behind the other is a round-the-world trip worth \$7,200. The producer of the game show has placed a subliminal message on the curtain containing the gag prize, which makes the probability of choosing the gag prize, which makes the probability of choosing the gag prize, which makes the probability of choosing the gag prize, which makes the probability of choosing the gag prize, which makes the probability of choosing the gag prize, which makes the probability of choosing the gate prize value of the selection, and what is the standard deviation of that selection?

7.3 Expected returns: You have chosen biology as your college major because you would like to be a medical doctor. However, you find that the probability of being accepted to medical school is about 10 percent. If you are accepted to medical school, then your starting salary when you gradu-

INTERMEDIATE ≥ 7.13 Expected returns: José is thinking about purchasing a soft drink machine and placing it in a business office. He knows that there is a 5 percent probability that someone who walks by the machine will make a purchase from the machine, and he knows that the profit on each soft drink sold is \$0.10. If José expects at housand people per day to pass by the machine and requires a complete return of his investment in one year, then what is the maximum price that he should be willing to pay for the soft drink machine? Assume 250 working days in a year, and ignore taxes and the time value of money.
 7.14 Interpreting the variance and standard deviation: The distribution of grades in an introductory finance class is normally distributed, with an expected grade of 75. If the standard deviation

EXCEL PROBLEMS

Nearly all problems can be solved using Excel templates at the student Web site within *WileyPLUS*. 7.27 David is going to purchase two stocks to form the initial holdings in his portfolio. Iron stock has a expected return of 15 percent, while Copper stock has an expected return of 20 percent. If David plans to invest 30 percent of his funds in Iron and the remainder in Copper, what will be the expected return from his portfolio? What if David invests 70 percent of his funds in Iron stock?
7.28 Peter knows that the covariance in the return on two assets is −0.0025. Without knowing the expected return of two assets, esplain what that covariance means.

CFA PROBLEMS ≥ 11.38 FITCO is considering the purchase of new equipment. The equipment costs \$350,000, and an additional \$110,000 is needed to install it. The equipment will be depreciated straight-line to zero over a five-year life. The equipment will generate additional annual revenues of \$265,000, and it will have annual cash operating expenses of \$83,000. The equipment will be sold for \$85,000 after five years. An inventory investment of \$73,000 is required during the life of the investment. FITCO is in the 40 percent tax bracket, and its cost of capital is 10 percent. What is the project NPPV? a. \$47,818. b. \$63,658.

CFA PROBLEMS

Problems from CFA readings are included in the Question and Problem section in appropriate chapters.

SAMPLE TEST PROBLEMS

Finally, five or more Sample Test Problems call for straightforward applications of the chapter concepts. These problems are intended to be representative of the kind of problems that may be used in a test, and instructors can encourage students to solve them as if they were taking a quiz. Solutions are provided in the Instructor's Manual.

c. \$80,189.**d.** \$97,449.

Sample Test Problems

7.1 Given the following information from Capstone Corporation, what price would the CAPM predict that the company's stock will trade for one year from today? Risk free rate: 3%

Market risk premium: 8%

- Beta: 0.65
- Current stock price: \$64.61 Annual dividend: \$1.92
- 7.2 You are considering investing in a mutual fund. The fund is expected to earn a return of 15 percent in the next year. If its annual return is normally distributed with a standard deviation of 6.5 percent, what return can you expect the fund to beat 95 percent of the time?
- 3.3 You have just invested in a portfolio of three stocks. The amount of money that you invested in each stock and its beta are summarized below. Calculate the beta of the portfolio and use the capital asset pricing model (CAPM) to compute the expected rate of return for the portfolio. Assume that the expected rate of return on the market is 15 percent and that the risk-free rate is 7 percent.

Investment \$200,000

Stock

А

• END OF PART ETHICS CASES

Ethics is an important topic in finance and this text addresses ethical issues in several ways. In Chapter 1, we introduce a framework for consideration of ethical issues in corporate finance. Many ethical issues can be analyzed in the context of informational asymmetry between parties to a transaction, conflicts of interest, breaches of confidentiality, and breaches of fiduciary duty (principalagent relationships); we highlight examples of such analysis throughout the text. In addition, seven ethics cases are included throughout this book in order to help students better understand how to analyze ethical dilemmas in the context of the framework. Real company examples are presented, including timeless cases about Arthur Anderson and Martha Stewart's scandal involving ImClone, and more timely topics such as the subprime mortgage crisis and the advent of sustainable living plans by corporations. Each case includes questions for followup discussion in class or as an assignment.

Beta

1.50

New to This Edition

In revising Fundamentals of Corporate Finance we have improved the presentation and organization of key topics, added important new content, updated the text to reflect changes in market and business conditions since the second edition was written, improved key in-chapter pedagogical features, and added to the number and quality of the end-of-chapter problem sets.

Improved Content, Presentation, and Organization

In preparing this edition of Fundamentals of Corporate Finance, we have extensively edited discussions throughout the text, rearranged the order of material, and added new content to improve the depth and effectiveness of the presentation. For example, we have moved the discussion of capital rationing from Chapter 12 to Chapter 10. This change improves the flow of the text by presenting capital rationing concepts along with other general capital budgeting concepts, rather than in a chapter that otherwise focuses only on evaluating the economics of individual projects. In addition, we have incorporated a more complete discussion of forms of business organization into Chapter 1. This more complete discussion was previously only found in Chapter 18, a chapter that not all instructors have time to cover in an introductory course. The change makes it easier for instructors to expose students to the increasingly varied forms of business organization that we are seeing in practice. We have also added a new section to Chapter 12 that introduces the concept of the economic break-even point. This new section helps students better understand how concepts that are presented earlier in the text can be used to assess the overall economic viability of a project.

Current Financial Market and Business Information

Throughout the text, all financial market and business information for which more current data are available have been updated. Not only have the exhibits been updated, but financial values such as interest rates, risk premia, and foreign currency exchange rates have been updated throughout the discussions in text, in-text examples, and end-of-chapter problems. In addition, all of the chapter opener vignettes have either been replaced or updated. Two of these examples are from 2012, 13 are from 2013, and six are from 2014. All of the chapter openers provide timely examples of how the material covered in the chapter is relevant to financial decision-making.

In-Chapter Features

The Learning Objectives at the beginning of each chapter have been revised to more fully reflect the important content in the associated sections of the chapters.

New **Building Intuition Boxes** have been added where appropriate and existing Building Intuition Boxes have been edited to ensure clarity.

All Learning by Doing Applications have been reviewed and, where appropriate, updated or replaced.

All Decision-Making Examples have been reviewed and updated where necessary.

The Summary of Learning Objectives and Key Equations at the end of each chapter have been updated to reflect changes in the chapter text and to improve the peda-gogical value of these features.

Refined and Extended Problem Sets

We have carefully edited the end-of-chapter questions and problems throughout the book to ensure that the examples are current and clearly presented. New Self-Study Problems, Discussion Questions, and Questions and Problems have been added to ensure appropriate coverage of key concepts at all levels of difficulty. In addition, more than a hundred new Sample Test Problems have been incorporated into the book in order to make these problems as representative of the key content in the book as possible. The total number of end-of-chapter questions and problems, including self-study problems and self-test questions, for the entire text has increased to 1,196.

Instructor and Student Resources

Fundamentals of Corporate Finance, Third Edition, features a full line of teaching and learning resources that were developed under the close review of the authors. Driven by the same basic beliefs as the textbook, these supplements provide a consistent and well-integrated learning system. This hands-on package guides *instructors* through the process of active learning and provides them with the tools to create an interactive learning environment. With its emphasis on activities, exercises, and the Internet, the package encourages *students* to take an active role in the course and prepares them for decision making in a real-world context.

WileyPLUS *WileyPLUS* with ORION is a research-based, online environment for effective teaching and learning. *WileyPLUS* builds students' confidence because it takes the guesswork out of studying by providing students with a clear roadmap: what to do, how to do it, if they did it right. This interactive approach focuses on:

Design: Research-based design is based on proven instructional methods. Content is organized into small, more accessible amounts of information, helping students build better time management skills.

Engagement: Students can visually track their progress as they move through the material at a pace that is right for them. Engaging in individualized self-quizzes followed by immediate feedback helps to sustain their motivation to learn.

Outcomes: Self-assessment lets students know the exact outcome of their effort at any time. Advanced reporting allows instructors to easily spot trends in the usage and performance data of their class in order to make more informed decisions.

With WileyPLUS, students will always know:

- What to do: Features, such as the course calendar, help students stay on track and manage their time more effectively.
- How to do it: Instant feedback and personalized learning plans are available 24/7.
- If they're doing it right: Self-evaluation tools take the guesswork out of studying and help students focus on the right materials.

WileyPLUS for *Fundamentals of Corporate Finance, Third Edition* includes numerous valuable resources, among them:

- Animated Learning by Doing Applications
- Wiley Corporate Finance Video Collection
- Prerequisite Course Reviews
- Animated Tutorials
- Excel Templates and Spreadsheet Solutions
- Flashcards and Crossword Puzzles
- Problem Walkthrough Videos
- Narrated PowerPoint Review
- Student Study Guide
- Hot Topics Modules
- Learning Styles Survey

Book Companion Site—For Instructors.

An extensive support package, including print and technology tools, helps you maximize your teaching effectiveness. We offer useful supplements for instructors with varying levels of experience and different instructional circumstances.

On this Web site instructors will find electronic versions of the Solutions Manual, Test Bank, Instructor's Manual, Computerized Test Bank, and other valuable resources: www.wiley.com/ college/Parrino.

Instructor's Manual. Included for each chapter are lecture outlines, a summary of learning objectives and key equations, and alternative approaches to the material. The *Solutions Manual* includes detailed solutions to the Before You Go On questions, Self-Study problems, Discussion Questions, and all of the Questions and Problems at the end of each chapter.

Test Bank. With over 2000 questions, the test bank allows instructors to tailor examinations according to study objectives and difficulty. Multiple-choice, true/false, and essay questions are included.

Computerized Test Bank. The computerized test bank allows instructors to create and print multiple versions of the

same test by scrambling the order of all questions found in the Word version of the test bank. The computerized test bank also allows users to customize exams by altering or adding new problems.

PowerPoint Presentations. The PowerPoint presentations contain a combination of key concepts, figures and tables, and problems and examples from the textbook as well as lecture notes and illustrations.

Book Companion Site — For Students.

The *Fundamentals of Corporate Finance* student Web site provides a wealth of support materials that will help students develop their conceptual understanding of class material and increase their ability to solve problems. On this Web site students will find Excel templates, study tools, web quizzing, and other resources: www.wiley.com/college/Parrino.

ACKNOWLEDGMENTS

The nearly 300 colleagues listed below provided valuable feedback during the development process and added greatly to the content and pedagogy of the program. Their commitment to teaching and willingness to become involved in such a project was a source of inspiration to the authors. We would like to acknowledge the contribution made by the following professors whose thoughtful comments contributed to the quality, relevancy, and accuracy of the Parrino Corporate Finance program.

Reviewers

Saul Adelman, Miami University Kenneth Ahern, University of Southern California Esther Ancel, University of Wisconsin-Milwaukee Ronald Anderson, Temple University Gene Andrusco, California State University San Bernardino Evrim Akdogu, Koç University Kofi Amoateng, North Carolina Central University Kavous Ardalan, Marist College Bala Arshanapalli, Indiana University Northwest Saul Auslander, Bridgewater State College Alan Bailey, University of Texas San Antonio Robert Balik, Western Michigan University John Banko, University of Florida Babu Baradwaj, Towson University Nina Baranchuk, University of Texas at Dallas Karen Barnhart, Missouri State University Janet Bartholow, Kent State University John Becker-Blease, Washington State University Omar Benkato, Ball State University Vashishta Bhaskar, Duquesne University Wilfred Jerome Bibbins, Troy University Hamdi Bilici, California State University Long Beach Ken Bishop, Florida Atlantic University David Blackwell, University of Kentucky Charles Blaylock, Murray State University

Vigdis Boasson, Central Michigan University Carol Boyer, Long Island University David Bourff, Boise State University Joe Brocato, Tarleton State University Jeffrey Brookman, Idaho State University Jeff Bruns, Bacone College James Buck, East Carolina University Juan Cabrera, Ramapo College Michael Carter, University of North Texas-Dallas Theodore Chadwick, Boston University Surya Chelikani, Quinnipiac University Ji Chen, University of Colorado Denver Jun Chen, University of North Carolina at Charlotte Yea-Mow Chen, San Francisco State University Paul Chiou, Shippensburg University William Chittenden, Texas State University Tarun Chordia, Emory University Ting-Heng Chu, East Tennessee State University Cetin Ciner, University of North Carolina at Wilmington Jonathan Clarke, Georgia Tech Thomas Coe, Quinnipiac University Hugh Colaco, Aston University Colene Coldwell, Baylor University Boyd D. Collier, Tarleton State University Roger Collier, Northeastern State University Lary B. Cowart, The University of Alabama at Birmingham Susan J. Crain, Missouri State University Tony Crawford, University of Montana Sandeep Dahiva, Georgetown University Julie Dahlquist, University of Texas at San Antonio Brent Dalrymple, University of Central Florida Amadeu DaSilva, California State University, Fullerton Sergio Davalos, University of Washington Diane Del Guercio, University of Oregon Zane Dennick-Ream, Robert Morris University John Dexter, Northwood University Robert Dildine, Metropolitan State University Robert Dubil, University of Utah Heidi Dybevik, University of Iowa Michael Dyer, University of Illinois at Urbana-Champaign David Eckmann, University of Miami Susan Edwards, Grand Valley State University Ahmed El-Shahat, Bradley University Frank Elston, University of South Queensland Maryellen Epplin, University of Central Oklahoma Stephen Ferris, University of Missouri Columbia Ron Filante, Pace University J. Howard Finch, Samford University Kathy Fogel, Suffolk University Joann Fredrickson, Bemidji State University Sharon Garrison, University of Arizona Louis Gasper, University of Dallas John Gawryk, Central Michigan University Edward Graham, University of North Carolina at Wilmington Richard P. Gregory, East Tennessee State University Nicolas Gressis, Wright State University Anthony Gu, SUNY Geneseo Roxane Gunser, University of Wisconsin Platteville

Manak Gupta, Temple University Sally Guyton, Texas A&M University Matthew Haertzen, Northern Arizona University Karen Hallows, University of Maryland Karen Hamilton, Georgia Southern University John Hatem, Georgia Southern University George Haushalter, Pennsylvania State University Andrew Head, Western Kentucky University Matthew Hood, University of Southern Mississippi James Howard, University of Maryland University College Jian Huang, Towson University Christy Huebner Caridi, Marist College Stephen Huffman, University of Wisconsin Oshkosh Rob Hull, Washburn University Kenneth Hunsader, University of South Alabama Jae-Kwang Hwang, Virginia State University Zahid Iqbal, Texas Southern University Jide Iwawere, Howard University Benjamas Jirasakuldech, Slippery Rock University Surendranath Jory, University of Sussex Jarl Kallberg, Thunderbird School of Global Management Ahmet Karagozoglu, Hofstra University Burhan Kawosa, Wright State University Gary Kayakachoian, University of Rhode Island Ayla Kayhan, Louisianna State University Baton Rouge James Kehr, Miami University of Ohio Peppi Kenny, Western Illinois University James Keys, Florida International University Robert Kieschnick, University of Texas Dallas Jaemin Kim, San Diego State University Kee Kim, Missouri State University Kenneth Kim, SUNY Buffalo Brett King, University of North Alabama Halil Kiymaz, Rollins College John Knight, University of the Pacific C.R. Krishna-Swamy, Western Michigan University Robert Krell, University of Phoenix Thomas J. Krissek, Northeastern Illinois University Raman Kumar, Virginia Tech University George Kutner, Marquette University Frances Kwansa, University of Delaware Julia Kwok, Northeastern State University Pamela LaBorde, Western Washington University Stephen Lacewell, Murray State University Gene Lai, Washington State University Mark Laplante, University of Wisconsin-Madison Duong Le, University of Arkansas at Little Rock Jerry Leabman, Bentley College Gregory La Blanc, University of California Berkeley Rick Le Compte, Wichita State University Alice Lee, San Francisco State University Cheng Few Lee, Rutgers University Jeong Lee, University of North Dakota Richard Lee, Barton College Canlin Li, The Federal Reserve System Mingsheng Li, Bowling Green State University Bing Liang, University of Massachusetts Amherst

Wendell Licon, Arizona State University Tempe Steven Lifland, High Point University Ralph Lim, Sacred Heart University Bingxuan Lin, University of Rhode Island Hong-Jen Lin, Brooklyn College Jason Lin, Truman State University David Lins, University of Illinois Peter Locke, Texas Christian University Robert Lutz, University of Utah Yulong Ma, California State University Long Beach Y. Lal Mahajan, Monmouth University Dana Manner, University of Miami Carol Mannino, Milwaukee School of Engineering Timothy Manuel, University of Montana Barry Marchman, Georgia Tech Richard Mark, Dowling College Brian Maris, Northern Arizona University Rand Martin, Bloomsburg University of Pennsylvania Richmond Matthews, University of Maryland Leslie Mathis, University of Memphis Stefano Mazzotta, Kennesaw State University Joseph McCarthy, Bryant University Lee McClain, Western Washington University Michael McNamara, Washington State University Kathleen McNichol, La Salle University Seyed Mehdian, University of Michigan at Flint Robert Meiselas, St. John's University Timothy Michael, University of Houston Clear Lake Jill Misuraca, University of Tampa Sunil Mohanty, University of St. Thomas Dianne Morrison, University of Wisconsin, La Crosse Shane Moser, University of Mississippi Michael Muoghalu, Pittsburg State University Suzan Murphy, University of Tennessee Dina Naples-Layish, SUNY Binghamton Vivian Nazar, Ferris State University Steven Nenninger, Sam Houstin State University Chee Ng, Fairleigh Dickinson University Brian Nichols, Missouri Southern State University Terry D. Nixon, Miami University Deniz Ozenbas, Montclair State University Vivek Pandey, University of Texas Tyler James Pandjiris, University of Missouri at St. Louis Nick Panepinto, Flagler College Coleen Pantalone, Northeastern University Robert Pavlik, Elon University Ivelina Pavlova, University of Houston, Clear Lake, Anil Pawar, San Diego State University Janet Payne, Texas State University Chien-Chih Peng, Morehead State University G. Michael Phillips, California State University Northridge James Philpot, Missouri State University Greg Pierce, Pennsylvania State University Steve Pilloff, George Mason University Wendy Pirie, CFA Institute Tony Plath, University of North Carolina, Charlotte Vassilis Polimenis, Aristotle University of Thessaloniki

Percy Poon, University of Nevada Las Vegas Terry Pope, Abilene Christian University Gary Powell, Queens University of Charlotte Richard Powell, Villanova University Dev Prasad, University of Massachusetts Lowell Rose Prasad, Central Michigan University Shoba Prekumar, Iowa State University Robert Puelz, Southern Methodist University Russell B. Raimer, Cleveland State University S. Rao, University of Louisiana at Lafayette Vadhindran K. Rao, Metropolitan State University Jong Rhim, University of Southern Indiana Greg Richey, California State University, San Bernadino Hong Rim, Shippensburg University Luis Rivera, Arcadia University Kenneth Roskelley, Mississippi State University Bruce Rubin, Old Dominion University David Rystrom, Western Washington University Helen Saar, Dixie State University Murray Sabrin, Ramapo College Sundarrajan Sankar, Tarleton State University Mukunthan Santhanakrishnan, Idaho State University Salil Sarkar, University of Texas Arlington Steven Scheff, Florida Gulf Coast University George Seldat, Southwest Minnesota State University Vivek Sharma, University of Michigan Dearborn Val Sibilkov, University of Wisconsin, Milwaukee Ana Silva, Merrimack College Betty Simkins, Oklahoma State University Stillwater Sudhir Singh, Frostburg State University Kim Small, Illinois State University Joe Smolira, Belmont University Andrew Spieler, Hofstra University Cliff Stephens, Louisiana State University Mark Hoven Stohs, California State University, Fullerton Gene Stout, Central Michigan University David Suk, Rider University Charlene Sullivan, Purdue University Michael Sullivan, University of Nevada at Las Vegas Alice Sun, California State Polytechnic University Pomona Srinivasan Sundaram, Ball State University Janikan Supanvanij, St. Cloud State University Robert Sweeney, Wright State University Philip Swensen, Summit Capital Investment Group Chu-Sheng Tai, Texas Southern University Thomas Tallerico, Stony Brook University Richard Taylor, Arkansas State University Bill Templeton, Butler University Diana Tempski, University of Wisconsin, La Crosse Rhonda Tenkku, University of Missouri at St. Louis Manish Tewari, The College at Brockport, State University of New York Philip Thames, California State University Long Beach Harold Thiewes Minnesota State University Paul Thistle, University of Nevada at Las Vegas Nancy Titus-Piersma, Oklahoma State at Stillwater Damir Tokic, ESC Rennes School of Business

K.C. Tseng, California State University Fresno Sergey Tsyplakov, University of South Carolina David Tufte, Southern Utah University Cathyann Tully, Kean University Marian Turac, College of Charleston Harry Turtle, Washington State University Pieter Vandenberg, San Diego State University Bonnie Van Ness, University of Mississippi Kristen Van Rensselaer, University of North Alabama Zoë L. Van Schyndel, Evergreen State College Sanjay Varshney, California State University Sacramento Rahul Verma, University of Houston Downtown Jill Lynn Vihtelic, Saint Mary's College Anna Vygodina, California State University Sacramento Jonathan Wagoner, Fairmont State University Joe Walker, University of Alabama at Birmingham Peggy Ward, Wichita State University Jim Washam, Arkansas State University Gwendolyn Webb, Baruch College Paul Weinstock, Ohio State University Susan White, University of Maryland Michael Willoughby, University of California San Diego Yangru Wu, Rutgers Business School Feixue Xie, University of Texas at El Paso Yan (Alice) Xie, University of Michigan Dearborn Devrim Yaman, Western Michigan University Hua Yang, Old Dominion University David Yamoah, Kean University Keven Yost, Auburn University Lei Yu, University of Notre Dame Jasmine Yur-Austin, California State University, Long Beach David Zhang, Arizona State University West Ge Zhang, William Patterson University Zhong-guo Zhou, California State University Northridge Kermit C. Zieg, Florida Institute of Technology

Advisory Board

Francisca M. Beer, California State University, San Bernardino Susan J. Crain, Missouri State University Praveen Kumar Das, University of Louisiana Amadeu DaSilva, California State University, Fullerton Beverly Hadaway, The University of Texas at Austin James D. Keys, Florida International University G. Michael Phillips, California State University, Northridge Helen Saar, Dixie State University Mark Stohs, California State University, Fullerton Devrim Yaman, Western Michigan University Jasmine Yur-Austin, California State University, Long Beach Kermit C. Zieg, Florida Institute of Technology

Class Testers

Saul W. Adelman, Miami University Babu Baradwaj, Towson University Jeffrey Brookman, Idaho State University Juan Cabrera, Ramapo College Michael Carter, University of North Texas Jun Chen, University of North Carolina at Charlotte Jonathan Clarke, Georgia Tech Thomas Coe, Quinnipiac College Hugh Colaco, Aston University Boyd Collier, Tarleton State University Roger Collier, Northeastern State University James Cordiero, The College at Brockport, State University of New York Irving DeGraw, St. Petersburg Junior College Zane Dennick-Ream, Robert Morris College Michael Dyer, University of Illinois, Urbana-Champaign Ronald Filante, Pace University Louis Gasper, University of Dallas J. Edward Graham, University of North Carolina, Wilmington Richard Gregory, East Tennessee State University Nicolas Gressis, Wright State University Matthew Haertzen, Northern Arizona University Karen L. Hamilton, Georgia Southern University Gary Kayakachoian, University of Rhode Island Peppi Kenny, Western Illinois University Julia Kwok, Northeastern State University Pamela LaBorde, Western Washington University Duong Le, University of Arkansas at Little Rock Mingsheng Li, Bowling Green State University Wendell Licon, Arizona State University Steven Lifland, High Point University Jason Lin, Truman State University Hong-Jen Lin, Brooklyn College Robert Lutz, University of Utah Brian A. Maris, Northern Arizona University Suzan Murphy, University of Tennessee S. P. UMA Rao, University of Louisiana at Lafavette Luis Rivera, Arcadia University David Rystrom, Western Washington University Murray Sabrin, Ramapo College Michael Sullivan, University of Nevada at Las Vegas Chu-Sheng Tai, Texas Southern University Thomas Tallerico, Stony Brook University Diana Tempski, University of Wisconsin, La Crosse Jonathan Wagoner, Fairmont State College

Accuracy Checkers

Robert J. Balik, Western Michigan University Babu Baradwaj, Towson University Jim P. DeMello, Western Michigan University Sharon Garrison, University of Arizona Sanjay Jain, Salem State University Mary Lou Poloskey, The University of Texas at Austin Inayat U. Mangla, Western Michigan University Jill Misuraca, University of Tampa Dianne Morrison, University of Wisconsin Napoleon Overton, University of Memphis Craig A. Peterson, Western Michigan University James Philpot, Missouri State University Judith Swisher, Western Michigan University Rhonda Tenkku, University of Missouri

The following people developed and revised valuable student and instructor resources available on the book companion site and *WileyPLUS*:

Babu Baradwaj, Towson University Charles Beauchamp, Middle Tennessee State University Michael Carter, University of North Texas, Dallas James DeMello, Western Michigan University James Dow, California State University, Northridge James Keys, Florida International University Patrick Lach, Eastern Illinois University Pamela LaBorde, Western Washington University Wendell Licon, Arizona State University Steven Lifland, High Point University Bridget Lyons, Sacred Heart University G. Michael Phillips, California State University, Northridge Dianne Morrison, University of Wisconsin, La Crosse Mary Lou Poloskey, The University of Texas at Austin Timothy Sargent, California State University, Northridge Philip Thames, California State University, Long Beach Susan White, University of Maryland Devrim Yaman, Western Michigan University

Contributor Team

We owe a special thanks to members of the contributor team for their hard work, exceptional creativity, consummate communications skills, and advice: Dr. Babu Baradwaj of Towson State University and Dr. Wendell Licon of Arizona State University who wrote the Instructor's Manual, student Study Guide, and major sections of the end-of-chapter materials for the First Edition. Dr. Norm Bowie of the University of Minnesota wrote most of the ethics cases. Petra Kubalova, of the Schwan Food Company, worked extensively with Professor Kidwell on this project during and after completing her MBA studies at Georgetown University. Dr. Zekiye Selvili of University of Southern California, Steven Gallaher of Southern New Hampshire University, and Nicholas Crane of University of Texas at Austin also contributed in a number of areas.

Publishing Team

We also thank the Wiley publishing team for always being calm, supportive, and gracious under fire as we suffered the travails of college textbook writing and revising, where deadlines are always yesterday. Those showing extraordinary patience and support include Joseph Heider, Senior Vice President and General Manager; Timothy Stookesberry, Vice President and Editorial Director; George Hoffman, Vice President and Executive Publisher; and Susan Elbe, Vice President, Market Development & Marketing. Those warranting special praise are Jennifer Manias, Project Editor, who coordinated the complex scheduling of the book and all of its resources; Joel Hollenbeck, Executive Editor; Courtney Luzzi, Assistant Editor; Tai Harriss, Senior Editorial Assistant; Amy Scholz, Director of Marketing; Karolina Zarychta, Senior Marketing Manager; and the Wiley sales force for their creativity and success in selling our book. Other Wiley staff who contributed to the text and media include Barbara Heaney, Director, Product and Market Development; Mia Brady, Marketing Assistant; Allie K. Morris, Senior Product Designer; Greg Chaput, Product Designer; Elena Santa Maria, Senior Media Specialist; Dorothy Sinclair, Content Manager; Valerie Vargas, Senior Production Editor; MaryAnn Price, Senior Photo Editor; Amanda Bustard, Photo Researcher; Harry Nolan, Design Director; and Maureen Eide, Senior Designer.

Colleagues

Robert Parrino would also like to thank some of his colleagues for their inspiration and helpful discussions. Among those who have significantly influenced this book are Robert Bruner of University of Virginia, Jay Hartzell of University of Texas at Austin, and Mark Huson of University of Alberta. Special thanks are owed to Clifford Smith, of University of Rochester, whose classes helped Dr. Parrino make sense of finance. In addition, recognition should go to Michael J. Barclay, who inspired generations of students through his selfless support and example and who was both a great researcher and teacher.

David Kidwell would like to thank some of his former professors and colleagues for their inspiration and willingness to share their intellectual capital. George Kaufman and Michael Hopewell who contributed to Dr. Kidwell's knowledge of economics and finance and were critical professors during his doctoral program at the University of Oregon. Richard West, former dean and professor at the University of Oregon, who unlocked the secrets of research and inspired Dr. Kidwell's love of teaching and scholarship. Robert Johnson of Purdue University whose dignity and academic bearing served as an academic role model and whose love of teaching and research inspired Dr. Kidwell to follow in his footsteps and to write this book. David Blackwell of Texas A&M University who helped conceptualize the idea for the book and contributed numerous insights to various chapters during the book's writing. Finally, to John Harbell and Jonas Mittelman, of San Francisco State University who started Dr. Kidwell on his academic journey.

Thomas Bates would like to thank the professors, coauthors, and colleagues who have made him a better scholar and educator, including Robert Williams, who first introduced him to the analytical elegance of economics, and Kenneth Lehn who inspired him to pursue excellence in research and teaching in the field of finance.

Brief Contents

PART 1 INTRODUCTION

1 The Financial Manager and the Firm 1

PART 2 FOUNDATIONS

- 2 The Financial System and the Level of Interest Rates 26
- Financial Statements, Cash Flows, and Taxes 50
- 4 Analyzing Financial Statements 83

PART 3 VALUATION OF FUTURE CASH FLOWS AND RISK

- 5 The Time Value of Money 127
- **6** Discounted Cash Flows and Valuation 160
- 7 Risk and Return 199
- 8 Bond Valuation and the Structure of Interest Rates 237
- 9 Stock Valuation 269

PART 4 CAPITAL BUDGETING DECISIONS

- 10 The Fundamentals of Capital Budgeting 300
- 11 Cash Flows and Capital Budgeting 345

- 12 Evaluating Project Economics 384
- 13 The Cost of Capital 411

PART 5 WORKING CAPITAL MANAGEMENT AND FINANCING DECISIONS

- 14 Working Capital Management 443
- 15 How Firms Raise Capital 474
- 16 Capital Structure Policy 506
- 17 Dividends, Stock Repurchases, and Payout Policy 549

PART 6 BUSINESS FORMATION, VALUATION, AND FINANCIAL PLANNING

- 18 Business Formation, Growth, and Valuation 573
- 19 Financial Planning and Managing Growth 609

PART 7 OPTIONS IN CORPORATE FINANCE AND INTERNATIONAL DECISIONS

- 20 Options and Corporate Finance 644
- 21 International Financial Management 673

Contents

PART 1 INTRODUCTION

CHAPTER 1 The Financial Manager and the Firm 1

- 1.1 THE ROLE OF THE FINANCIAL MANAGER 2
 - Stakeholders 2 It's All About Cash Flows 2
 Building Intuition: Cash Flows Matter Most to Investors 4 Three Fundamental Decisions in Financial
 Management 4 Building Intuition: Sound Investments Are Those Where the Value of the Benefits
 Exceeds Their Cost 5 Building Intuition: Financing
 Decisions Affect the Value of the Firm 6
- FORMS OF BUSINESS ORGANIZATION 6
 Sole Proprietorships 6 Partnerships 8
 Corporations 8
- **1.3** MANAGING THE FINANCIAL FUNCTION 10

Organizational Structure 10 Positions Reporting to the CFO 10 External Auditor 11 The Audit Committee 11 The Compliance and Ethics Director 12

- 1.4 THE GOAL OF THE FIRM 12
 - What Should Management Maximize? 12 Why Not Maximize Profits? 12

Building Intuition: The Timing of Cash Flows Affects Their Value 13 Building Intuition: The Riskiness of Cash Flows Affects Their Value 13 Maximize the Value of the Firm's Stock 13 Building Intuition: The Financial Manager's Goal Is to Maximize the Value of the Firm's Stock 14 Can Management Decisions Affect Stock Prices? 14

1.5 AGENCY CONFLICTS: SEPARATION OF OWNERSHIP AND CONTROL 15

> Ownership and Control 15 Agency Relationships 15 Do Managers Really Want to Maximize Stock Price? 16 Aligning the Interests of Managers and Stockholders 16 Sarbanes-Oxley and Other Regulatory Reforms 18

1.6 THE IMPORTANCE OF ETHICS IN BUSINESS 20

Business Ethics20Are Business Ethics Differentfrom Everyday Ethics?20Types of Ethical Conflictsin Business20The Importance of an Ethical Businessness Culture21Serious Consequences22

Summary of Learning Objectives • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

PART 2 FOUNDATIONS

CHAPTER 2

The Financial System and the Level of Interest Rates 26

2.1 THE FINANCIAL SYSTEM 27

The Financial System at Work28through the Financial System28

How Funds Flow

2.2 DIRECT FINANCING 29

A Direct Market Transaction 30 Investment Banks and Direct Financing 30

2.3 TYPES OF FINANCIAL MARKETS 32

Primary and Secondary Markets 32 Exchanges and Over-the-Counter Markets 33 Money and Capital Markets 33 Public and Private Markets 33 Futures and Options Markets 34 2.4 MARKET EFFICIENCY 35

Efficient Market Hypotheses 35

2.5 FINANCIAL INSTITUTIONS AND INDIRECT FINANCING 36

Indirect Market Transactions 37 Financial Institutions and Their Services 37 Corporations and the Financial System 38

2.6 THE DETERMINANTS OF INTEREST RATE LEVELS 40

The Real Rate of Interest40Loan Contractsand Inflation42The Fisher Equation andInflation42Cyclical and Long-Term Trendsin Interest Rates44

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 3Financial Statements, Cash Flows, and Taxes 50

3.1 FINANCIAL STATEMENTS AND ACCOUNTING PRINCIPLES 51

The Annual Report51Generally AcceptedAccounting Principles52Fundamental AccountingPrinciples52International GAAP53Company: Diaz Manufacturing53

3.2 THE BALANCE SHEET 54

Current Assets and Liabilities 55 Long-Term Assets and Liabilities 56 Equity 57

3.3 MARKET VALUE VERSUS BOOK VALUE 59

A More Informative Balance Sheet 59 A Market-Value Balance Sheet 60

3.4 THE INCOME STATEMENT AND THE STATEMENT OF RETAINED EARNINGS 62

The Income Statement 62 The Statement of Retained Earnings 64

- **3.5** THE STATEMENT OF CASH FLOWS 65 Sources and Uses of Cash 65
- **3.6** TYING TOGETHER THE FINANCIAL STATEMENTS 68
- 3.7 CASH FLOWS TO INVESTORS 69

Net Income versus the Cash Flow to Investors69Cash Flow to Investors: Putting It All Together72

3.8 FEDERAL INCOME TAX 73

Corporate Income Tax Rates 74 Average versus Marginal Tax Rates 74 Unequal Treatment of Dividends and Interest Payments 75

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 4 Analyzing Financial Statements 83

4.1 BACKGROUND FOR FINANCIAL STATEMENT ANALYSIS 84

Perspectives on Financial Statement Analysis 84 Guidelines for Financial Statement Analysis 85

4.2 COMMON-SIZE FINANCIAL STATEMENTS 86

Common-Size Balance Sheets 86 Common-Size Income Statements 87

4.3 FINANCIAL RATIOS AND FIRM PERFORMANCE 88

Why Ratios Are Better Measures88Short-TermLiquidity Ratios89Efficiency Ratios91Leverage Ratios95Profitability Ratios99Market-Value Indicators102ConcludingComments on Ratios103

4.4 THE DUPONT SYSTEM: A DIAGNOSTIC TOOL 103

An Overview of the DuPont System 103 The ROA Equation 104 The ROE Equation 105 The DuPont Equation 106 Applying the DuPont System 106 Is Maximizing ROE an Appropriate Goal? 107

4.5 SELECTING A BENCHMARK 108

Trend Analysis 108 Industry Analysis 108 Peer Group Analysis 109

4.6 USING FINANCIAL RATIOS 110

Performance Analysis of Diaz Manufacturing 110 Limitations of Financial Statement Analysis 113

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

ETHICS CASE: A Sad Tale: The Demise of Arthur Andersen 125

PART 3 VALUATION OF FUTURE CASH FLOWS AND RISK

CHAPTER 5 The Time Value of Money 127

5.1 THE TIME VALUE OF MONEY 128

Consuming Today or Tomorrow 128 Building Intuition: The Value of Money Changes with Time 129 Time Lines as Aids to Problem Solving 129 Financial Calculator 129

5.2 FUTURE VALUE AND COMPOUNDING 130

Single-Period Investment 130 Two-Period Investment 131 The Future Value Equation 132 The Future Value Factor 134 **Future Value** Factor Tables 134 Applying the Future Value Formula 134 Building Intuition: Compounding Drives Much of the Earnings on Long-Term Investments 136 Calculator Tips for Future Value Problems 138

5.3 PRESENT VALUE AND DISCOUNTING 142

Single-Period Investment 142 Multiple-Period Investment 142 The Present Value Equation 143 Future and Present Value Equations Are the Same 143 Applying the Present Value Formula 144 The Relations Among Time, the Discount Rate, and Present Value 146 Calculator Tips for Present Value Problems 146 Future Value Versus Present Value 147

5.4 ADDITIONAL CONCEPTS AND APPLICATIONS 148

Finding the Interest Rate149Finding How ManyPeriods It Takes an Investment to Grow a CertainAmount150The Rule of 72151CompoundGrowth Rates151Concluding Comments153

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 6 Discounted Cash Flows and Valuation 160

6.1 MULTIPLE CASH FLOWS 161

Future Value of Multiple Cash Flows161PresentValue of Multiple Cash Flows164

6.2 LEVEL CASH FLOWS: ANNUITIES AND PERPETUITIES 168

Present Value of an Annuity 168 Future Value of an Annuity 177 Perpetuities 179 Annuities Due 180

6.3 CASH FLOWS THAT GROW AT A CONSTANT RATE 182

Growing Annuity 182 Growing Perpetuity 183

6.4 THE EFFECTIVE ANNUAL INTEREST RATE 184 Why the Confusion? 184 Calculating the Effective Annual Interest Rate 185 Comparing Interest Rates 185 Consumer Protection Acts and Interest Rate Disclosure 187 The Appropriate Interest Rate Factor 187

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

Appendix: Deriving the Formula for the Present Value of an Ordinary Annuity 196 Problem 196 ETHICS CASE: Buy It on Credit and Be True to Your School 197

CHAPTER 7 Risk and Return 199

- 7.1 RISK AND RETURN 200 Building Intuition: More Risk Means a Higher Expected Return 201
- 7.2 QUANTITATIVE MEASURES OF RETURN 201
 Holding Period Returns 201 Expected Returns 202
- 7.3 VARIANCE AND STANDARD DEVIATION AS MEASURES OF RISK 206

Calculating Variance and Standard Deviation 206 Interpreting Variance and Standard Deviation 207 Historical Market Performance 210

7.4 RISK AND DIVERSIFICATION 213

Single-Asset Portfolios214Portfolios with Morethan One Asset216Building Intuition: DiversifiedPortfolios Are Less Risky222The Limits of Diversificationfication222

7.5 SYSTEMATIC RISK 223

Why Systematic Risk Is All That Matters223Building Intuition: Systematic Risk Is the Risk ThatMatters223Measuring Systematic Risk223

- 7.6 COMPENSATION FOR BEARING SYSTEMATIC RISK 226
- 7.7 THE CAPITAL ASSET PRICING MODEL 227

The Security Market Line 228 The Capital Asset Pricing Model and Portfolio Returns 229

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 8Bond Valuation and the Structure of Interest Rates 237

8.1 CORPORATE BONDS 238

Market for Corporate Bonds 238 Bond Price Information 239 Types of Corporate Bonds 239

8.2 BOND VALUATION 240

The Bond Valuation Formula241Calculator Tip:Bond Valuation Problems242Par, Premium,and Discount Bonds243SemiannualCompounding245Zero Coupon Bonds245

8.3 BOND YIELDS 247

Yield to Maturity 248 Effective Annual Yield 249 Realized Yield 251

8.4 INTEREST RATE RISK 251

Bond Theorems 251 Bond Theorem Applications 253

8.5 THE STRUCTURE OF INTEREST RATES 254

Marketability 254 Call Provision 255 Default Risk 255 The Term Structure of Interest Rates 257

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

ETHICS CASE: The Subprime Mortgage Market Meltdown: How Did It Happen? 266

CHAPTER 9 Stock Valuation 269

9.1 THE MARKET FOR STOCKS 270

Secondary Markets 270 Secondary Markets and Their Efficiency 271 Stock Market Indexes 273 Reading the Stock Market Listings 273 Common and Preferred Stock 274 Preferred Stock: Debt or Equity? 274

9.2 COMMON STOCK VALUATION 275

A One-Period Model 276 A Perpetuity Model 277 The General Dividend Valuation Model 278 The Growth Stock Pricing Paradox 279

9.3 STOCK VALUATION: SOME SIMPLIFYING ASSUMPTIONS 280

Zero-Growth Dividend Model 280 Constant-Growth Dividend Model 280 Computing Future Stock Prices 283 The Relation between *R* and *g* 285 Mixed (Supernormal) Growth Dividend Model 285

9.4 VALUING PREFERRED STOCK 289

Preferred Stock with a Fixed Maturity 289 Preferred Stock with No Maturity 290

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

ETHICS CASE: Insider Trading: Have I Got a Stock Tip for You! 298

PART 4 CAPITAL BUDGETING DECISIONS

CHAPTER 10 The Fundamentals of Capital Budgeting 300

10.1 AN INTRODUCTION TO CAPITAL BUDGETING 301

The Importance of Capital Budgeting301The CapitalBudgeting Process301Sources of Information302Classification of Investment Projects303BasicCapital Budgeting Terms303Building Intuition:Investment Decisions Have Opportunity Costs304

10.2 NET PRESENT VALUE 304

Valuation of Real Assets 304 NPV—The Basic Concept 305 NPV and Value Creation 305 Framework for Calculating NPV 306 Net Present Value Techniques 308 Concluding Comments on NPV 311

10.3 THE PAYBACK PERIOD 312

Computing the Payback Period312How thePayback PeriodPerforms314Discounted PaybackPeriod315Evaluating the Payback Rule316

10.4 THE ACCOUNTING RATE OF RETURN 317

10.5 INTERNAL RATE OF RETURN 317

Calculating the IRR 318 When the IRR and NPV Methods Agree 320 When the NPV and IRR Methods Disagree 321 Modified Internal Rate of Return (MIRR) 324 IRR versus NPV: A Final Comment 326

10.6 INVESTMENT DECISIONS WITH CAPITAL RATIONING 327

Capital Rationing in a Single Period 328 Capital Rationing Across Multiple Periods 330

10.7 CAPITAL BUDGETING IN PRACTICE 331Practitioners' Methods of Choice 331Postaudit and Periodic Reviews 331

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 11 Cash Flows and Capital Budgeting 345

11.1 CALCULATING PROJECT CASH FLOWS 346

Building Intuition: Capital Budgeting Is ForwardLooking 346Incremental After-Tax Free CashFlows 346The FCF Calculation 347Building Intuition: Incremental After-Tax Free CashFlows Are What Stockholders Care About In CapitalBudgeting 348Cash Flows from Operations 349Cash Flows Associated with Capital Expenditures andNet Working Capital 349The FCF Calculation: AnExample 350FCF versus Accounting Earnings 353

11.2 ESTIMATING CASH FLOWS IN PRACTICE 354

Five General Rules for Incremental After-Tax Free Cash Flow Calculations 354 Nominal versus Real Cash Flows 357 Tax Rates and Depreciation 359 Computing the Terminal-Year FCF 363 Expected Cash Flows 365 Building Intuition: We Discount *Expected* Cash Flows in an NPV Analysis 366

11.3 FORECASTING FREE CASH FLOWS 367

Cash Flows from Operations 367 Cash Flows Associated with Capital Expenditures and Net Working Capital 368

11.4 SPECIAL CASES (OPTIONAL) 369

Projects with Different Lives 369 When to Replace an Existing Asset 370 The Cost of Using an Existing Asset 372 When to Harvest an Asset 373

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

ETHICS CASE: Unilever's Sustainable Living Plan 382

CHAPTER 12 Evaluating Project Economics 384

12.1 VARIABLE COSTS, FIXED COSTS, AND PROJECT RISK 385

Cost Structure and Sensitivity of EBITDA to Revenue Changes 386 Cost Structure and Sensitivity of EBIT to Revenue Changes 388 Building Intuition: High Fixed Costs Mean Larger Fluctuations in Cash Flows and Profits 388

12.2 CALCULATING OPERATING LEVERAGE 392

Degree of Pretax Cash Flow Operating Leverage 392 Degree of Accounting Operating Leverage 393 Building Intuition: Revenue Changes Drive Profit Volatility Through Operating Leverage 393

12.3 BREAK-EVEN ANALYSIS 394

Pretax Operating Cash Flow Break-Even 394 Accounting Operating Profit (EBIT) Break-Even 396

12.4 THE ECONOMIC BREAK-EVEN POINT 398

12.5 RISK ANALYSIS 401

Sensitivity Analysis401Scenario Analysis402Simulation Analysis403

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 13The Cost of Capital 411

13.1 THE FIRM'S OVERALL COST OF CAPITAL 412

The Finance Balance Sheet413Building Intuition:The Market Value of a Firm's AssetsEquals the MarketValue of the Claims on Those Assets414HowFirms Estimate Their Cost of Capital414BuildingIntuition: A Firm's Cost of Capital Is a WeightedAverage of All of Its Financing Costs416

13.2 THE COST OF DEBT 417

Key Concepts for Estimating the Cost of Debt 417 Building Intuition: The Current Cost of Long-Term Debt Is What Matters When Calculating WACC 418 Estimating the Current Cost of a Bond or an Outstanding Loan 418 Taxes and the Cost of Debt 419 Estimating the Cost of Debt for a Firm 420

13.3 THE COST OF EQUITY 422

Common Stock 422 Preferred Stock 428

13.4 USING THE WACC IN PRACTICE 430

Calculating WACC: An Example 430 Limitations of WACC as a Discount Rate for Evaluating Projects 432 Alternatives to Using WACC for Evaluating Projects 435

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

PART 5 WORKING CAPITAL MANAGEMENT AND FINANCING DECISIONS

CHAPTER 14 Working Capital Management 443

14.1 WORKING CAPITAL BASICS 444

Working Capital Terms and Concepts444WorkingCapital Accounts and Trade-Offs446

- 14.2 THE OPERATING AND CASH CONVERSION CYCLES 447
 - Operating Cycle 448 Cash Conversion Cycle 450
- **14.3** WORKING CAPITAL MANAGEMENT STRATEGIES 452

Flexible Current Asset Management Strategy452Restrictive Current Asset Management Strategy452The Working Capital Trade-Off452

14.4 ACCOUNTS RECEIVABLE 454

Terms of Sale 454 Aging Accounts Receivable 455

14.5 INVENTORY MANAGEMENT 457 Economic Order Quantity 457

Just-in-Time Inventory Management 458

- 14.6 CASH MANAGEMENT AND BUDGETING 458 Reasons for Holding Cash 458 Cash Collection 459
- 14.7 FINANCING WORKING CAPITAL 460

Strategies for Financing Working Capital 460 Financing Working Capital in Practice 462 Sources of Short-Term Financing 462

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 15 How Firms Raise Capital 474

15.1 BOOTSTRAPPING 475

How New Businesses Get Started 475 Initial Funding of the Firm 476

15.2 VENTURE CAPITAL 476

The Venture Capital Industry476Why VentureCapital Funding Is Different477The VentureCapital Funding Cycle478Venture CapitalistsProvide More Than Financing480The Cost ofVenture Capital Funding481

15.3 INITIAL PUBLIC OFFERING 482

Advantages and Disadvantages of Going Public 482 Building Intuition: Investors View Seasoned Securities as Less Risky Than Unseasoned Securities 483 Investment Banking Services 483 Origination 483 Underwriting 484 Distribution 485 The Proceeds 486

15.4 IPO PRICING AND COST 487

The Underpricing Debate487IPOs Are Consis-tently Underpriced488The Cost of an IPO489

15.5 GENERAL CASH OFFER BY A PUBLIC COMPANY 491

Competitive versus Negotiated Sale 491 The Cost of a General Cash Offer 493

15.6 PRIVATE MARKETS AND BANK LOANS 494

Private versus Public Markets 494 Private Placements 495 Private Equity Firms 495 Private Investments in Public Equity 496 Commercial Bank Lending 497 Concluding Comments on Funding the Firm 499

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

ETHICS CASE: Profiting from Death: "Janitor's Insurance" 504

CHAPTER 16 Capital Structure Policy 506

16.1 CAPITAL STRUCTURE AND FIRM VALUE 507

The Optimal Capital Structure 507 Building Intuition: The Optimal Capital Structure Minimizes the Cost of Financing a Firm's Activities 507 The Modigliani and Miller Propositions 508 Building Intuition: Capital Structure Choices Do Not Affect Firm Value If They Do Not Affect the Value of the Free Cash Flows to Investors 509 Building Intuition: The Cost of Equity Increases With Financial Leverage 512

16.2 THE BENEFITS AND COSTS OF USING DEBT 516 The Benefits of Debt 516 The Costs of Debt 522 Building Intuition: People Behave Differently Toward a Firm in Financial Distress, and This Increases Bankruptcy Costs 524

16.3 TWO THEORIES OF CAPITAL STRUCTURE 528 The Trade-Off Theory 528 The Pecking Order Theory 529 The Empirical Evidence 529

16.4 PRACTICAL CONSIDERATIONS IN CHOOSING A CAPITAL STRUCTURE 531

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems •

APPENDIX: LEASING 538

CHAPTER 17 Dividends, Stock Repurchases, and Payout Policy 549

17.1 DIVIDENDS 550

Building Intuition: Dividends Reduce the Stockholders' Investment in a Firm 550 Types of Dividends 551 The Dividend Payment Process 552 Building Intuition: Dividend Announcements Send Signals to Investors 553

17.2 STOCK REPURCHASES 555

How Stock Repurchases Differ from Dividends 555 How Stock Is Repurchased 557

17.3 DIVIDENDS AND FIRM VALUE 558

Benefits and Costs of Dividends 559 Stock Price Reactions to Dividend Announcements 561 Dividends versus Stock Repurchases 562

17.4 STOCK DIVIDENDS AND STOCK SPLITS 564 Stock Dividends 564 Stock Splits 565

Reasons for Stock Dividends and Splits 565

17.5 SETTING A DIVIDEND PAYOUT 566

What Managers Tell Us 566 Practical Considerations in Setting a Dividend Payout 567

Summary of Learning Objectives • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

PART 6 BUSINESS FORMATION, VALUATION, AND FINANCIAL PLANNING

CHAPTER 18 Business Formation, Growth, and Valuation 573

18.1 STARTING A BUSINESS 574

Making the Decision to Proceed 574 Choosing the Right Organizational Form 575 Financial Considerations 577

18.2 THE ROLE OF THE BUSINESS PLAN 581

Why Business Plans Are Important 582 The Key Elements of a Business Plan 582

18.3 VALUING A BUSINESS 583

Fundamental Business Valuation Principles 583 Building Intuition: The Value of a Business Is Specific to a Point in Time 584 Building Intuition: The Value of a Business Is Not the Same to All Investors 584 Business Valuation Approaches 584

18.4 IMPORTANT ISSUES IN VALUATION 598

Public versus Private Companies 598 Young (Rapidly Growing) versus Mature Companies 599 Controlling Interest versus Minority Interest 599 Key People 600

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

CHAPTER 19 Financial Planning and Managing Growth 609

19.1 FINANCIAL PLANNING 610

The Planning Documents611Building Intuition:A Firm's Strategy Drives Its Business Decisions612Concluding Comments613

19.2 FINANCIAL PLANNING MODELS 613

The Sales Forecast614Building a Financial PlanningModel614A Simple Planning Model615

19.3 A BETTER FINANCIAL PLANNING MODEL 619

The Blackwell Sales Company619The IncomeStatement619The Balance Sheet620The Preliminary Pro Forma Balance Sheet622The Final Pro Forma Balance Sheet624

- **19.4** BEYOND THE BASIC PLANNING MODELS 626 Improving Financial Planning Models 626
- 19.5 MANAGING AND FINANCING GROWTH 628
 - External Funding Needed 628 A Graphical View of Growth 631 The Sustainable Growth Rate 632 Growth Rates and Profits 634 Growth as a Planning Goal 634

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

PART 7 OPTIONS IN CORPORATE FINANCE AND INTERNATIONAL DECISIONS

chapter 20

Options and Corporate Finance 644

20.1 FINANCIAL OPTIONS 645

Call Options 645 Put Options 647 American, European, and Bermudan Options 648 More on the Shapes of Option Payoff Functions 648 Building Intuition: Payoff functions for options are not linear 648

20.2 OPTION VALUATION 649

Limits on Option Values 649 Variables That Affect Option Values 651 The Binomial Option Pricing Model 652 Put-Call Parity 655 Valuing Options Associated with the Financial Securities That Firms Issue 656

20.3 REAL OPTIONS 657

Options to Defer Investment 658 Options to Make Follow-On Investments 658 Options to Change Operations 659 Options to Abandon Projects 660 Concluding Comments on NPV Analysis and Real Options 660

20.4 AGENCY COSTS 661

Agency Costs of Debt 661 Agency Costs of Equity 663

20.5 OPTIONS AND RISK MANAGEMENT 664

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

ETHICS CASE: Compensation—How Much Is Enough? 671

APPENDIX A

Future Value and Present Value Tables 703

APPENDIX ${f B}$ Solutions to Selected Questions and Problems 713

CHAPTER 21

International Financial Management 673

21.1 INTRODUCTION TO INTERNATIONAL FINANCIAL MANAGEMENT 674

Globalization of the World Economy674The Riseof Multinational Corporations675Factors AffectingInternational Financial Management676Goals ofInternational Financial Management677BasicPrinciples Remain the Same678Building Intuition:The Basic Principles of Finance ApplyNo Matter WhereYou Do Business678

21.2 FOREIGN EXCHANGE MARKETS 679

Market Structure and Major Participants679Foreign Exchange Rates679The EquilibriumExchange Rate681Foreign CurrencyQuotations682

21.3 INTERNATIONAL CAPITAL BUDGETING 687

Determining Cash Flows 687 Exchange Rate Risk 688 Country Risk 688 The Barcelona Example 688

21.4 GLOBAL MONEY AND CAPITAL MARKETS 691

The Emergence of the Euromarkets691TheEurocurrency Market691The EurocreditMarket692International Bond Markets692

21.5 INTERNATIONAL BANKING 694

Risks Involved in International Bank Lending 694 Eurocredit Bank Loans 695

Summary of Learning Objectives • Summary of Key Equations • Self-Study Problems • Solutions to Self-Study Problems • Discussion Questions • Questions and Problems • Sample Test Problems

Glossary 723

Subject Index 731

Company Index 747

THIRD EDITION

FUNDAMENTALS OF CORPORATE FINANCE



The Financial Manager and the Firm

© Estate of Stephen Laurence Strathdee/iStockphoto

n May 24, 2012, Apollo Global Management, LLC, announced the completion of a \$7.15 billion purchase of El Paso Corporation's oil and gas exploration and production business. This business, which would become an independent company named EP Energy, had drilling rights on almost 3.3 million acres of land with oil and gas reserves at varying levels of development. EP Energy had been acquired by Apollo and several other investors through a leveraged buyout—a transaction in which the purchaser uses a lot of debt to help pay for the acquisition.

Why did the investors decide to purchase El Paso Corporation's exploration and production business, and how did the El Paso managers and the investment group arrive at the \$7.15 billion price tag? Surely the investors did not plan to lose money. Apollo, a very successful investor in the energy sector, believed that EP Energy was an excellent investment even at a cost of \$7.15 billion. Sam Oh, a partner at Apollo, stated, "We are delighted to partner with one of North America's leading exploration and production franchises led by a world-class team of

managers." By taking advantage of the operational experience of the management team and by using a great deal of debt financing, EP Energy was expected to earn high returns for its investors.

Investors in leveraged buyouts like the EP Energy transaction use many of the concepts covered in this chapter and elsewhere in this book to create the most value possible. Managers of leveraged buyout firms like Apollo are paid in a way that provides them with strong incentives to focus on value creation. They create value by investing in companies only when the benefits exceed the cost, managing the assets of the companies they buy as efficiently as possible, and financing those companies with the least expensive combination of debt and equity. This chapter introduces you to the key financial aspects of these activities, and the remainder of the book fills in many of the details.

..... Learning Objectives

 Identify the key financial decisions facing the financial manager of any business.

- Identify common forms of business organization in the United States and their respective strengths and weaknesses.
- Describe the typical organization of the financial function in a large corporation.
- Explain why maximizing the value of the firm's stock is the appropriate goal for management.
- Discuss how agency conflicts affect the goal of maximizing stockholder value.
- Explain why ethics is an appropriate topic in the study of corporate finance.

CHAPTER PREVIEW

This book provides an introduction to corporate finance. In it we focus on the responsibilities of the financial manager, who oversees the accounting and treasury functions and sets the overall financial strategy for the firm. We pay special attention to the financial manager's role as a decision maker. To that end, we emphasize the mastery of fundamental financial concepts and tools which are used to make sound financial decisions that create value for stockholders. These financial concepts and tools apply not only to business organizations but also to other venues, such as government entities, not-for-profit organizations, and sometimes even our own personal finances. We open this chapter by discussing the three major types of decisions that a financial manager makes. We then describe common forms of business organization. After next discussing the major responsibilities of the financial manager, we explain why maximizing the value of the firm's stock is an appropriate goal for a financial manager. We go on to describe the conflicts of interest that can arise between stockholders and managers and the mechanisms that help align the interests of these two groups. Finally, we discuss the importance of ethical conduct in business.

1.1 THE ROLE OF THE FINANCIAL MANAGER

LEARNING OBJECTIVE

wealth the economic value of the assets someone possesses

stakeholder anyone other than an owner (stockholder) with a claim on the cash flows of a firm, including employees, suppliers, creditors, and the government

productive assets the long-term tangible and intangible assets a firm uses to generate cash flows The financial manager is responsible for making decisions that are in the best interests of the firm's owners, whether the firm is a start-up business with a single owner or a billion-dollar corporation owned by thousands of stockholders. The decisions made by the financial manager or owner should be one and the same. In most situations this means that the financial manager should make decisions that maximize the value of the owners' stock. This helps maximize the owners' **wealth**, which is the economic value of the assets the owner possesses. Our underlying assumption in this book is that most people who invest in businesses do so because they want to increase their wealth. In the following discussion, we describe the responsibilities of the financial manager in a new business in order to illustrate the types of decisions that such a manager makes.

Stakeholders

Before we discuss the new business, you may want to look at Exhibit 1.1, which shows the cash flows between a firm and its owners (in a corporation, the stockholders) and various stake-holders. A **stakeholder** is someone other than an owner who has a claim on the cash flows of the firm: *managers*, who want to be paid salaries and performance bonuses; *other employees*, who want to be paid wages; *suppliers*, who want to be paid for goods or services they provide; *creditors*, who want to be paid interest and principal; and the *government*, which wants the firm to pay taxes. Stakeholders may have interests that differ from those of the owners. When this is the case, they may exert pressure on management to make decisions that benefit them. We will return to these types of conflicts of interest later in the book. For now, though, we are primarily concerned with the overall flow of cash between the firm and its stockholders and stakeholders.

It's All About Cash Flows

To produce its products or services, a new firm needs to acquire a variety of assets. Most will be long-term assets, which are also known as **productive assets**. Productive assets can be tangible assets, such as equipment, machinery, or a manufacturing facility, or intangible assets, such as patents, trademarks, technical expertise, or other types of intellectual capital. Regardless of the type of asset, the firm tries to select assets that will generate the greatest cash flows for the firm's owners. The decision-making process through which the firm purchases productive assets is called *capital budgeting*, and it is one of the most important decision processes in a firm.



EXHIBIT 1.1 Cash Flows Between the Firm and Its Stakeholders and Owners (Stockholders)

3

A. Making business decisions is all about cash flows, because only cash can be used to pay bills and buy new assets. Cash initially flows into the firm as a result of the sale of goods or services. These cash inflows are used in a number of ways: to pay wages and salaries, to buy supplies, to repay creditors, and to pay taxes. **B.** Any cash that is left over (residual cash flows) can be reinvested in the business or paid as dividends to stockholders.

Once the managers of a firm have selected the firm's productive assets, they must raise money to pay for them. *Financing decisions* determine the ways in which firms obtain and manage long-term financing to acquire and support their productive assets. There are two basic sources of funds: debt and equity. Every firm has some equity because equity represents ownership in the firm. It consists of capital contributions by the owners plus cash flows that have been reinvested in the firm. In addition, most firms borrow from a bank or issue some type of long-term debt to finance productive assets.

After the productive assets have been purchased and the business is operating, the managers of the firm will try to produce products at the lowest possible cost while maintaining quality. This means buying raw materials at the lowest possible cost, holding production and labor costs down, keeping management and administrative costs to a minimum, and seeing that shipping and delivery costs are competitive. In addition, day-to-day finances must be managed so that the firm will have sufficient cash on hand to pay salaries, purchase supplies, maintain inventories, pay taxes, and cover the myriad of other expenses necessary to run a business. The management of current assets, such as money owed by customers who purchase on credit, inventory, and current liabilities, such as money owed to suppliers, is called *working capital management*.¹

A firm generates cash flows by selling the goods and services it produces. A firm is successful when these cash inflows exceed the cash outflows needed to pay operating expenses, creditors, and taxes. After meeting these obligations, managers of the firm can pay the remaining cash, called **residual cash flows**, to the owners as a cash dividend, or reinvest the cash in the business. The reinvestment of residual cash flows in the business to buy more productive assets is a very important concept. If these funds are invested wisely, they provide the foundation for the firm to grow and provide larger residual cash flows in the future for the owners. The reinvestment of cash flows (earnings) is the most fundamental way that businesses grow in size. Exhibit 1.1 illustrates how the revenue generated by productive assets ultimately becomes residual cash flows.

residual cash flows the cash remaining after a firm has paid operating expenses and what it owes creditors and in taxes; can be paid to the owners as a cash dividend or reinvested in the business

¹From accounting, *current assets* are assets that will be converted into cash within one year, and *current liabilities* are liabilities that must be paid within one year.

BUILDING Intuition

CASH FLOWS MATTER MOST TO INVESTORS

Intuition Cash is what investors ultimately care about when making an investment. The value of any asset—a share of stock, a bond, or a business—is determined by the cash flows it is expected to generate in the future. To understand this concept, just consider how much you would pay for an asset from which you could never expect to obtain any cash flows. Buying such an asset would be like giving your money away. It would have a value of exactly zero. Conversely, as the expected cash flows from an investment increase, you would be willing to pay more and more for it. A firm is unprofitable when it fails to generate sufficient cash inflows to pay operating expenses, creditors, and taxes. Firms that are unprofitable over time will be forced into **bankruptcy** by their creditors if the owners do not shut them down first. In bankruptcy the company will be reorganized or the company's assets will be liquidated, whichever is more valuable. If the company is liquidated, creditors are paid in a priority order according to the structure of the firm's financial contracts and prevailing bankruptcy law. If anything is left after all creditor and tax claims have been satisfied, which usually does not happen, the remaining cash, or residual value, is distributed to the owners.

bankruptcy legally declared inability of an individual or a company to pay its creditors

Three Fundamental Decisions in Financial Management

Based on our discussion so far, we can see that financial managers are concerned with three fundamental decisions when running a business:

- 1. Capital budgeting decisions: Identifying the productive assets the firm should buy.
- 2. Financing decisions: Determining how the firm should finance or pay for assets.
- **3.** Working capital management decisions: Determining how day-to-day financial matters should be managed so that the firm can pay its bills, and how surplus cash should be invested.

Exhibit 1.2 shows the impact of each decision on the firm's balance sheet. We briefly introduce each decision here and discuss them in greater detail in later chapters.

Deleves Chest

EX	ΗB	IT 1	.2

How the Financial Manager's Decisions Affect the Balance Sheet

Financial managers are concerned with three fundamental types of decisions: capital budgeting decisions, financing decisions, and working capital management decisions. Each type of decision has a direct and important effect on the firm's balance sheet and, ultimately, the success or failure of the firm.

Balance Sneet			
Assets		Liabilities and Equity	
Current assets (including cash, inventory, and	Working capital — management decisions deal with day-to-day financial matters and affect current assets, current liabilities, and net working capital.	Current liabilities (including short-term debt and accounts payable)	
accounts receivable)	Net working capital—the difference between current assets and current liabilities		
Long-term assets (productive assets, which may be tangible or intangible)	Capital budgeting decisions determine what long-term (productive) assets the firm will purchase.	Long-term debt (debt with a maturity of over one year)	
	Financing decisions determine the firm's <i>capital structure</i> —the combination of long-term debt and equity that will be used to finance the firm's long-term productive assets and net working capital.	Stockholders' equity	

Capital Budgeting Decisions

A firm's capital budget is simply a list of the productive (capital) assets that management wants to purchase over a budget cycle, typically one year. The capital budgeting decision process addresses which productive assets the firm should purchase and how much money the firm can afford to spend. As shown in Exhibit 1.2, capital budgeting decisions affect the asset side of the balance sheet and are concerned with a firm's long-term investments. Capital budgeting decisions, as we mentioned earlier, are among management's most important decisions. Over the long run, they have a large impact on the firm's success or failure. The reason is twofold. First, capital (productive) assets generate most of the cash flows for the firm. Second, capital assets are long term in nature. Once they are purchased, the firm owns them for a long time, and they may be hard to sell without taking a financial loss.

The fundamental question in capital budgeting is this: Which productive assets should the firm purchase? A capital budgeting decision may be as simple as a movie theater's decision to buy a popcorn machine or as complicated as Boeing's decision to invest more than \$6 billion to design and build the 787 *Dreamliner* passenger jet. Capital investments may also involve the purchase of an entire business, such as Google's purchase of the mobile software company Bump in May 2013.

Regardless of the project, a good capital investment is one in which the benefits are worth more to the firm than the cost of the asset. For example, in August 2012 Verizon Communications received government approval to purchase wireless spectrum (the right to use radio frequencies over which voice and data signals are sent) from a group of cable companies for \$3.9 billion. Presumably, Verizon expects that the investment will produce a stream of cash flows worth more than that. Suppose Verizon estimates that in terms of the current market value, the future cash flows from the wireless spectrum purchase are worth \$5 billion. Is the purchase a good deal for Verizon? The answer is yes because the value of the expected cash flow benefits from the purchase exceeds the cost by \$1.1 billion (\$5.0 billion – \$3.9 billion = \$1.1 billion). If the purchase of the wireless spectrum works out as planned, the value of Verizon will be increased by \$1.1 billion!

Not all investment decisions are successful. Just open the business section of any newspaper on any day, and you will find stories of bad decisions. For example, Universal Picture's 2009 comedy *Land of the Lost* reportedly cost over \$140 million in production and advertising expenses, but made only \$69.5 million in worldwide box office receipts. Even with U.S. DVD sales of approximately \$18 million, the overall cash flows from sales of the movie did not come close to covering its upfront costs. When, as in this case, the cost exceeds the value of the future cash flows, the project will decrease the value of the firm by that amount.

SOUND INVESTMENTS ARE THOSE WHERE THE VALUE OF THE BENEFITS EXCEEDS THEIR COST

Financial managers should invest in a capital project

only if the value of its future cash flows exceeds the cost of the project (benefits > cost). Such investments increase the value of the firm and thus increase stockholders' (owners') wealth. This rule holds whether you're making the decision to purchase new machinery, build a new plant, or buy an entire business.

BUILDING

Intuition

Financing Decisions

Financing decisions determine how firms raise cash to pay for their investments, as shown in Exhibit 1.2. Productive assets, which are long term in nature, are financed by long-term borrowing, equity investment, or both. Financing decisions involve trade-offs between the advantages and disadvantages of these financing alternatives for the firm.

A major advantage of debt financing is that debt payments are tax deductible for many corporations. However, debt financing increases a firm's risk because it creates a contractual obligation to make periodic interest payments and, at maturity, to repay the amount that is borrowed. Contractual obligations must be paid regardless of the firm's operating cash flow, even if the firm suffers a financial loss. If the firm fails to make payments as promised, it defaults on its debt obligation and could be forced into bankruptcy.

In contrast, equity has no maturity, and there are no guaranteed payments to equity investors. In a corporation, the board of directors has the right to decide whether dividends should be paid to stockholders. This means that if a dividend payment is reduced or omitted altogether,

BUILDING Intuition

FINANCING DECISIONS AFFECT THE VALUE OF THE FIRM

How a firm is financed with debt and equity affects the value of the firm. The reason is that the mix between debt and equity affects the taxes the firm pays and the probability that the firm will go bankrupt. The financial manager's goal is to determine the combination of debt and equity that minimizes the cost of financing the firm. the firm will not be in default. Unlike interest payments, however, dividend payments to stockholders are not tax deductible.

The mix of debt and equity on the balance sheet is known as a firm's **capital structure**. The term capital structure is used because long-term funds are considered capital, and these funds are raised in **capital markets**—financial markets where equity and debt instruments with maturities greater than one year are traded.

capital structure the mix of debt and equity that is used to finance a firm

capital markets

financial markets where equity and debt instruments with maturities greater than one year are traded

net working capital the dollar difference between total current assets and total current liabilities

Working Capital Management Decisions

Management must also decide how to manage the firm's current assets, such as cash, inventory, and accounts receivable, as well as its current liabilities, such as trade credit and accounts payable. The dollar difference between a firm's total current assets and its total current liabilities is called its **net working capital**, as shown in Exhibit 1.2. As mentioned earlier, working capital management is the day-to-day management of the firm's short-term assets and liabilities. The goals of managing working capital are to ensure that the firm has enough cash to pay its bills and invest any spare cash to earn interest.

The mismanagement of working capital can cause a firm to default on its debt and go into bankruptcy, even though, over the long term, the firm may be profitable. For example, a firm that makes sales to customers on credit but is not diligent about collecting the accounts receivable can quickly find itself without enough cash to pay its bills. If this condition becomes chronic, creditors can force the firm into bankruptcy if the firm is not able to obtain alternative financing.

A firm's profitability can also be affected by its inventory level. If the firm has more inventory than it needs to meet customer demands, it has too much capital tied up in assets that are not earning cash. Conversely, if the firm holds too little inventory, it can lose sales because it does not have products to sell when customers want them. Management must therefore determine the optimal inventory level.

BEFORE YOU GO ON

- 1. What are the three basic types of financial decisions managers must make?
- 2. Explain why you would make an investment if the value of the expected cash flows exceeds the cost of the project.
- **3.** Why are capital budgeting decisions among the most important decisions in the life of a firm?

1.2 FORMS OF BUSINESS ORGANIZATION

2 LEARNING OBJECTIVE

Firms are organized in a number of different ways in the United States. In this section we discuss some of the more common forms of organization and the factors that business owners consider when they choose which to use. Exhibit 1.3 summarizes key characteristics of common forms of business organization.

Sole Proprietorships

sole proprietorship a business owned by a single person A **sole proprietorship** is a business that is owned by a single person. Its life is limited to the period that the owner (proprietor) is associated with the business because there is no ownership interest that can be transferred to someone else—there is no stock or other such interest that can

EXHIBIT 1.3 Characteristics of Different Forms of Business Organization

Choosing the appropriate form of business organization is an important step in starting a business. This exhibit compares key characteristics of the most popular forms of business organization in the United States.

	Sole	Partnership		Corporation		Limited Liability
	Proprietorship	General	Limited	S-Corp.	C-Corp.	Company (LLC)
Cost to establish	Inexpensive	More costly	More costly	More costly	More costly	More costly
Life of entity	Limited	Flexible	Flexible	Indefinite	Indefinite	Flexible
Control by founder over business decisions	Complete	Shared	Shared	Depends on ownership	Depends on ownership	Shared
Access to capital	Very limited	Limited	Less limited	Less limited	Excellent	Less limited
Cost to transfer ownership	High	High	High	High	Can be low	High
Separation of management and investment	No	No	Yes	Yes	Yes	Yes
Potential owner/ manager conflicts	No	No	Some	Potentially high	Potentially high	Some
Ability to provide incentives to attract and retain high- quality employees	Limited	Good	Good	Good	Good	Good
Liability of owners	Unlimited	Unlimited	Unlimited for general partner	Limited	Limited	Limited
Tax treatment of income	Flow-through	Flow-through	Flow-through	Flow-through	Double tax	As elected
Tax deductibility of owner benefits	Limited	Limited	Limited	Limited	Less limited	Limited

be sold. A sole proprietorship ceases to exist when the proprietor stops being involved with the business. Many small businesses in the United States are organized this way.

A sole proprietorship is the simplest and least expensive form of business to set up and is the least regulated. To start a sole proprietorship, all you have to do is obtain the business licenses required by your local and state governments.

The ownership structure of a sole proprietorship has both advantages and disadvantages. Among the advantages is the fact that the proprietor does not have to share decision-making authority with anyone and can run the business as he or she chooses.

There are several disadvantages related to the fact that there is no stock or other ownership interest to sell. First, the amount of equity capital that can be raised to finance the business is limited to the owner's personal wealth. This can restrict growth for the business unless the proprietor is very wealthy. Second, it can be more costly to transfer ownership. The proprietor must sell the assets of the business directly, rather than indirectly through the sale of an ownership interest in an operating business. The business must essentially be re-established every time it is sold. Third, because the proprietor provides all of the equity capital and manages the business, there is no separation of the management and investment roles. This limits the ability of good managers to form a business if they do not also have capital to invest. Fourth, it is not possible to provide employees with compensation in the form of ownership interests, such as stock or stock options, which can help motivate them to work harder.

Another disadvantage of a sole proprietorship is that proprietor faces *unlimited liability*. If someone is harmed by the business, the proprietor's liability extends beyond the money invested in the business. The proprietor can lose some or all of his or her personal wealth too.

Finally, profits from a sole proprietorship *flow through* to the sole proprietor's personal tax return, meaning that the business does not pay taxes before profits are distributed to the owner. Because the business is not subject to income taxes, profits are not subject to double-taxation as is a C-corporation (described later). There are limitations on tax deductions for personal expenses, such as those associated with health insurance, but the costs of these limitations are often outweighed by the benefits from the flow-through of profits in a sole proprietorship.

Partnerships

A **partnership** consists of two or more owners who have joined together legally to manage a business. To form a partnership, the owners (partners) enter into an agreement that details how much capital each partner will invest in the partnership, what their management roles will be, how key management decisions will be made, how the profits will be divided, and how ownership will be transferred in case of specified events, such as the retirement or death of a partner. A *general partnership* is a partnership in which all of the partners are owners of (investors in) the business and active in managing it. In contrast, a *limited partnership* has both *general partners*, who are owners and managers, and *limited partners*, who are owners, but not managers.

Partnerships are more costly to form than sole proprietorships because the partners must hire an attorney to draw up and maintain the *partnership agreement*, which specifies the nature of the relationships between or among the partners. On the other hand, partnership agreements can be amended to allow for the business to continue when a partner leaves. The ability to make the life of a business independent of that of the partners increases the liquidity of the ownership interests, making it easier to raise capital and less costly for the partners to sell their interests at an attractive price.

Many of the other advantages and disadvantages of a general partnership are similar to those of a sole proprietorship. A key disadvantage of a *general partnership* is that, like the proprietor in a sole proprietorship, all partners have unlimited liability. This liability can be even worse than in a sole proprietorship because a general partner can be held liable for all of the partnership's debts and other obligations, regardless of what proportion of the business he or she owns or how the debt or other obligations were incurred.

The problem of unlimited liability is avoided for some partners in a *limited partnership* because limited partners can generally only lose the amount of money that they have invested in the business. In a limited partnership one or more general partners have unlimited liability and actively manage the business, while each limited partner is liable for business obligations only up to the amount of capital he or she invested in the partnership. In other words, the limited partners have **limited liability**. To qualify for limited partner status, a partner cannot be actively engaged in managing the business.

limited liability the legal liability of an investor is limited to the amount of capital invested in the business

corporation

partnership

its profits

two or more owners who have

manage a business and share

joined together legally to

a legal entity formed and authorized under a state charter; in a legal sense, a corporation is a "person" distinct from its owners

Corporations

Most large businesses are organized as corporations. A **corporation** is a legal entity authorized under a state charter. In a legal sense, it is a "person" distinct from its owners. For example, corporations can sue and be sued, enter into contracts, borrow money, and own assets. They can also be general or limited partners in partnerships, and they can own stock in certain types of other corporations. Because a corporation is an entity that is distinct from its owners, it can have an indefinite life. Corporations hold the majority of all business assets and generate the majority of business revenues and profits in the United States. The owners of a corporation are its stockholders.

Starting a corporation is more costly than starting a sole proprietorship. For example, it requires writing articles of incorporation and by-laws that conform to the laws of the state of incorporation. These documents spell out the name of the corporation, its business purpose, its intended life span (unless explicitly stated otherwise, the life is indefinite), the amount of stock to be issued, and the number of directors and their responsibilities. Over the life of a successful business, these costs are not very important. However, to a cash-strapped entrepreneur, they can seem substantial.

On the other hand, the corporate form of organization has several advantages. For example, shares in a corporation can be sold to raise capital from investors who are not involved in the business. This can greatly increase the amount of capital that can be raised to fund the business.

Another major advantage of a corporation is that stockholders have limited liability for debts and other obligations. Owners of corporations have limited liability because corporations are *legal persons* that take actions in their own names, not in the names of individual owners.

An *S*-corporation is a form of corporation that can be used by private businesses that meet certain requirements. An S-corporation can have only one class of stock and cannot have more than one hundred stockholders or any stockholders that are corporations or nonresident alien investors. In contrast, a *C*-corporation, which is the form used by public corporations, does not face such limits.

While there are more restrictions on S-corporations, there are also advantages. Specifically, all profits earned by an S-corporation pass directly to the stockholders, just as they pass to a sole proprietor or the partners in a partnership. This means that no taxes are paid at the corporate level.

In contrast, a major disadvantage of a C-corporation is that it must pay taxes on the income it earns. If the corporation pays a cash dividend, the stockholders must also pay taxes on the dividends they receive. Thus, the owners of C-corporations are subject to double taxation first at the corporate level and then at the personal level when they receive dividends.

Corporations can be classified as public or private. Most large companies prefer to operate as public corporations because large amounts of capital can be raised in public markets at a relatively low cost. **Public markets**, such as the New York Stock Exchange (NYSE) and NASDAQ, are regulated by the Securities and Exchange Commission (SEC).²

In contrast, **privately held**, or **closely held**, **corporations** are owned by a small number of investors, and their shares are not traded publicly. When a corporation is first formed, the common stock is often held by a few investors, typically the founder, a small number of key managers, and financial backers. Over time, as the company grows in size and needs larger amounts of capital, management may decide that the company should "go public" in order to gain access to the public markets. Not all privately held corporations go public, however.

Limited Liability Partnerships and Companies

Historically, law firms, accounting firms, investment banks, and other professional groups were organized as sole proprietorships or partnerships. For partners in these firms, all income was taxed as personal income, and general partners had unlimited liability for all debts and other obligations of the firm. It was widely believed that in professional partnerships, such as those of attorneys, accountants, or physicians, the partners should be liable individually and collectively for the professional conduct of each partner. This structure gave the partners an incentive to monitor each other's professional conduct and discipline poorly performing partners, resulting in a higher quality of service and greater professional integrity. Financially, however, misconduct by one partner could result in disaster for the entire firm. For example, a physician found guilty of malpractice exposes every partner in the medical practice to financial liability, even if the others never treated the patient in question.

In the 1980s, because of sharp increases in the number of professional malpractice cases and large damages awards in the courts, professional groups began lobbying state legislators to create hybrid forms of business organization. One such hybrid is known as a **limited liability partnership (LLP)**. An LLP combines some of the limited liability characteristics of a corporation with the tax advantage of a partnership. While liability varies from state to state, LLP partners in general have more limited liability than general partners in regular partnerships. Typically, they are not personally liable for any other partner's malpractice or professional misconduct. Like regular partnerships, income to the partners of an LLP is taxed as personal income.

A **limited liability company (LLC)** is another hybrid form of organization that is becoming increasingly common. Like LLPs, LLCs have benefited founders of many businesses Visit the Web sites of the NYSE and NASDAQ at http://www.nyse.com and http://www.nasdaq.com to get more information about market activity.

public markets markets regulated by the Securities and Exchange Commission in which securities such as stocks and bonds are publicly traded

privately held, or closely held, corporations corporations whose stock is not traded in the public markets

limited liability partnership (LLP) and limited liability company (LLC) hybrid business organizations that combine some of the advantages of corporations and partnerships; in general, income to the partners is taxed only as personal income, but the partners have limited liability

²We examine the public and private markets in more detail in Chapters 2 and 15.